

This draft for discussion comprises selected extracts of a monograph under preparation. They include drafts of a couple of chapters in their entirety, the concluding sections of several chapters and in some cases, only the chapter headings.

AFRICAN DEVELOPMENT: DEAD ENDS AND NEW BEGINNINGS

Meles Zenawi

(The author is the Prime Minister of Ethiopia. The views expressed are personal and do not necessarily reflect the official position of the Government.).

Introduction

The political and economic renaissance of Africa is an issue that continues to preoccupy Africans' and non-Africans alike. Various methods of achieving such a renaissance have been proposed. Most of these proposals are variations of the dominant neo-liberal paradigm of development. My argument is that the neo-liberal paradigm is a dead end, is incapable of bringing about the African renaissance, and that a fundamental shift in paradigm is required to bring about the African renaissance.

The monograph is divided into three parts. Part one provides the theoretical basis of the argument and is divided into seven chapters. Chapter one briefly outlines the fundamentals of the neo-liberal political economy, identifies the flaws in the neo-liberal political economy and outlines the social underpinnings of a market economy. It argues that social development is essential for economic development and that social development cannot be brought about by market mechanisms alone.

Chapter two focuses on technological development which is at the heart of economic development. It shows the massive market failures associated with technological development, argues that technological capability accumulation is at the center of the development efforts of developing countries, and concludes that such development cannot be brought about by market mechanisms alone.

Chapter three deals with other market failures, with a special emphasis on capital markets. It argues that market failures are deep and pervasive in developing countries and that market mechanisms alone cannot bring about accelerated development.

Chapter four addresses the role of agriculture in development. The market failures that agricultural development faces are assessed and the relationship between equity and growth is investigated. Chapter five deals with the role of FDI in development.

Chapter 6 is concerned with the role of the state in accelerating growth and in addressing market failures. It argues that the state has historically played a crucial role in accelerating development and analyzes the nature of the state which can be most effective in addressing market failures and accelerating growth.

Chapter 7 deals with democracy and development. While there has been accelerated development without democracy, democratization has been an essential element of the vision of the African renaissance. It is argued that a developmental state can be a democratic state, and indeed that a democratic developmental state is likely to be more successful in its development efforts than others. The requirements for the emergence and evolution of democracy are also examined.

Part two (chapters 8 to 13) provides the practical experience to show the validity of the analysis in part one. The development experience of Taiwan and Korea is analyzed in some detail. The ways and means that the two governments used to address the various market failures are analyzed. The evolution of developmental states in the two countries is assessed. The internal and external circumstances that contributed to the success of their development effort are outlined. Part two provides specific and practical examples showing that government intervention by a "developmentalist", if not a full-fledged developmental state, is key to successful development.

Part three analyzes the current circumstances in Africa, argues that the neo-liberal reforms have failed and explains why the neo-liberal paradigm has failed and does so in 8 chapters (chapters 14-21).

Chapter 14 explains the nature and genesis of the predatory African state and how its internal logic and dynamics led Africa to a dead end and crisis by the mid-eighties. It analyzes the political economy of rent-seeking in Africa. Chapter 15 analyzes the successful development experience of Botswana and the political economy of Botswana. It shows how the different political economy of Botswana has been responsible both for the successes and challenges of Botswana.

Chapter 16 presents the neo-liberal analysis of the political and economic problems of Africa and the political and economic reform agenda proposed by it.

Chapter 17 and 18 present the results of the neo-liberal reforms. It is argued that the reforms have failed, that the reform programs have not taken Africa out of the dead end in which it found itself in the mid-eighties and that on the contrary the neo-liberal reforms have taken Africa to another dead end. It is suggested that is so because of the fundamental flaws of the neo-liberal paradigm.

Chapter 19 deals with the reform experience of some African countries. The last chapter argues for the need of a paradigm shift to bring about the African renaissance and assesses the possibilities of bringing about such a paradigm shift. Some of the essential steps that are needed to bring that about are also briefly outlined.

1. The Neo-Liberal Political Economy and Social Capital

(Extract from the chapter)

1.4 Conclusions

The neo-liberal paradigm which suggests a non-activist and non-interventionist state, a night watchman state as conducive to economic growth bases such conclusions on two pillars. One pillar has to do with the assertion that competitive markets are both pervasive and pareto efficient. The second pillar has to do with the neo-liberal political economy based on the theory of socially wasteful rent-seeking activities and the rational choice theory of solely self-interest maximizing individuals.

Government created rent does not necessarily have to be socially wasteful. It becomes wasteful only if solely self-interest maximizing individuals use it to create wealth at the expense of society and only if the state is incapable of improving on the market- i.e. there are no market failures.

The theory of solely self-interest maximizing individuals does not hold water. History, common everyday observation, and theoretical analysis based on the two suggests that an economy based on complex economic interaction such as a market economy requires a blend of self-interested and non-self interested behavior: a blend of social and individual norms that maximize survival potential within an appropriate social context. In the absence of such norms, the state, if it can exist as a coherent corporate entity for any period of time, becomes predatory. A properly behaved night watchman state populated by solely interest maximizing individuals is thus a practical and theoretical impossibility. Only individuals with a blend of self-interested and non-self interested behavior can create a night watchman state, and such people are equally capable of creating a state which intervenes in the economy in the larger interest of society.

Creating the proper blend of norms, values and rules to reduce uncertainty and transaction costs is a critical factor in accelerated growth and development. The creation of such social values and norms is called social development or social capital accumulation. Social development is thus not only an essential element of development but also a critical instrument of accelerated economic growth.

The accumulation of social capital, which plays such a critical role in accelerating economic growth, is a public good which has increasing returns to scale. It is, hence, undersupplied by the market and is subject to vicious and virtuous cycles. It is created by social activity by civic engagement in the context of horizontal and dense networks and inculcated and sustained through modeling, socialization and sanctions. The state plays a critical role in social capital accumulation through undermining patronage networks and promotion of fairness and equity, through the promotion of participation and democracy, and through appropriate sanctions and efforts at socialization.

2. The Black Box of Development

(Extract from Chapter)

2.5 Conclusions

The neo- liberal paradigm correctly identifies technological change as the heart of the development process and as the only source of continuous increase in per capita income. However, it treats technological change as an exogenous factor, as a sort of Black-box and thus does not even attempt to explain it in economic terms. Endogenous growth theory, which in many respects is a variant of neo-liberal theory does begin to open up the Black- box and begin to see the market failures involved, but fails to provide a comprehensive understanding of the process as it is tied to the most basic neo-liberal assumption of the pervasiveness and efficiency of markets.

Developing countries cannot compete simply on the basis of factor endowment, or by buying up the latest machinery. They need to assimilate technology developed elsewhere, and they need to continuously move up the technology ladder if they are to achieve continued growth and development. In other words technological capability accumulation is as central to developing countries as it is to developed countries. The difference is that in developing countries such accumulation takes place primarily through the assimilation of foreign technology rather the development of new technology.

Firms take a lot of risk and incur heavy expenses to identify and assimilate foreign technology but are unable to fully appropriate the benefits of their efforts. Indeed, national development would be hindered if they were to fully appropriate the benefit of their effort as the newly introduced technology would not be diffused. There is thus a fundamental market failure in technological capability accumulation in developing countries. Moreover, the required external environment is such that it would not be possible to create it through the market mechanism alone. Technological capability accumulation in developing countries is as plagued by pervasive market failure as is the development of new technology.

A deeper analysis of technological change shows that it is plagued by information failures and extreme forms of information asymmetry, of increasing returns, of extensive externalities, and coordination failures. It shows that technology has the essential characteristics of a public good. In other words, a deeper analysis of technological change shows that in both developed and developing countries technological change takes place in an environment of pervasive market failures.

A historical analysis of technological development shows that successful societies have developed national innovation systems that address market failures, and that such systems are based more on the structures and histories of the economies rather than on relative factor prices. It shows that there are differing national systems of innovation reflecting the differing histories and structures of the economies which can do the job, and that it would be impossible to envisage successful technological development outside of such institutional environment in any country, developed or developing.

Cleary then, when it comes to the heart of accelerated economic growth, when it comes to technological development, the neo-liberal assumption of efficient competitive markets has no basis in fact or theory. Its assumption of a night watchman state as an instrument that facilitates technological development by letting the markets do the job while it stands guard protecting property rights and enforcing contracts has no basis in fact, in history or theory. The neo-liberal paradigm cannot explain or guide the heart of development, technological growth.

Stiglitz goes even further and plausibly argues that the neo-liberal theory is fundamentally inconsistent with technological change. Where there are competitive markets and prices are equal to marginal costs, there is no room for technological change, where there is room for technological change there are no competitive markets.

3 Market Failure in Developing Countries

4. Agriculture as the Engine of Growth

(Extract from chapter)

4.6 Conclusions

The theoretical and historical analysis presented has shown that the engine of development in its initial phases is and has to be agriculture, and that this is so not only because growth in the relatively massive agricultural sector will have bigger impacts on total growth of the economy directly but also because agricultural growth accelerates non-agricultural growth and structural transformation even more. The neo-liberal school of thought is thus absolutely correct in taking agriculture as the engine of the early phases of development.

Market failures related to technology, capital markets, weak and absent markets is even more pronounced in agriculture than in the other sectors leading to a number of vicious circles and poverty traps. A night watchman state, a state whose intervention in the economy is very limited would be unable to overcome the vicious circles and poverty traps. The neo-liberal paradigms advocacy of such a state in developing countries is thus likely to keep such countries mired in poverty traps.

The neo-liberal school's focus on resource transfer as the key linkage between agriculture and non-agriculture is misplaced. Moreover, its focus on the price mechanism to carry-out this resource transfer is also misplaced. The neo-liberal assertion that getting prices right is the efficient way of carrying-out the transfer and that getting prices right means liberalizing agricultural products markets is fundamentally wrong. Getting prices right means getting the non-market determinants of agricultural prices right. Once the determinants of agricultural prices are right various price and non-price mechanisms can be used to carry-out the transfer without negatively affecting the incentives to farmers. If the determinants of agricultural prices (non-market determinants) are wrong, liberalization of agricultural prices will not ensure the right prices. The single minded focus on liberalization of agricultural prices at the expense of dealing with the real non-market determinants of these prices will lead to the continuation of the pervasive market failures in agriculture.

Our analysis supported by historical evidence shows that equitable distribution of assets in the rural areas plays a critical role in accelerating agricultural development and overall development and structural transformation. Equity accelerates the adoption and diffusion of agricultural technology. Equity plays a vital role in the establishment and strengthening of market support and other rural organizations and institutions. Equity enhances the linkage between agriculture and non-agriculture and increases the multiplier effect of agricultural growth on the growth of non-

agriculture and structural transformation. Equity facilitates the accumulation of social capital which in turn plays a critical role in accelerating development.

Contrary to popular perception the fundamental theory of the neo-liberal paradigm does not require that there be inequitable distribution of assets to ensure Pareto- efficiency. On the contrary the theory asserts that Pareto-efficiency is neutral vis-a-vis equity. You can have Pareto-efficiency in both equitable and non-equitable asset distribution environments. The neo-liberal assertion that inequality will have to increase during the development process and decrease later is not based on any theory but on an empirical assessment that the rich save more than the poor and that increased saving through inequitable distribution of wealth is crucial for accelerated growth.

The view that the savings rate of a country is higher when wealth distribution is skewed has been and can be contested empirically. But even if this were to be true there is no theoretical reason why piling up savings where social capital is low and opportunistic behavior, violence and unrest is prevalent, where technological diffusion and adoption is severely retarded, where market supporting institutions are absent or non-existent, and where the multiplier effect of agricultural growth on non-agricultural is severely curtailed will accelerate growth. It is not possible to convincingly argue that increasing savings has such an overwhelming impact on growth as to override all the retarding effects of inequitable distribution of wealth on development.

5. Foreign Investment and Development

6. The Developmental State

(Extract from chapter)

6.3 Conclusions

In previous chapters we have argued that inhibiting rent seeking, behavior does not depend on the size of the state on the degree of its activism in economic matters but on the nature of the state without however defining the nature of the state, which can be an activist state at the same time as inhibiting socially wasteful rent- seeking activity. Similarly we have argued that developing countries face formidable market failures and institutional inadequacies which create vicious circles and poverty traps, which can adequately be addressed only by an activist state. We have shown that the historical practice bears this out. We had not however defined what sort of an activist state is required. The analysis of the developmental state in this chapter is intended to fill this gap.

We can thus conclude that in the end, development is a political process first and economic and social process later. It is the creation of a political set-up that is conducive to accelerated development that sets the ball of development rolling. Only when there is a state that has the characteristics of a developmental state can one meaningfully discuss the elimination of rent-seeking behavior. In its absence rent-seeking will be rampant no matter what the size of the state might be. Only in the context of such a political environment can one debate about development policy in a meaningful manner. In its absence all government policy and action however limited and timid it might be will be riddled with rent-seeking behavior and this particularly so in developing countries as these countries will be coming out of a social and political environment where vertical, patron-client networks are pervasive.

The neo-liberal paradigm states that socially wasteful rent-seeking is the result of government activity and of the size of government activism. It does not distinguish between different types of state activism. This leads it to conclude that most if not all government intervention in the economy is detrimental to growth and hence to suggest that the night watchman state is the best state from the point of view of accelerated growth. Historical practice has shown that state intervention has been critical in the development process. Economic theory has shown that developing countries are riddled with vicious circles and poverty traps that can only be removed by state action. The theory of the developmental state completes the alternative paradigm by showing what type of state can intervene in the economy to accelerate growth while at the same time limiting socially wasteful rent-seeking activities.

7. Democracy, Developmental State and Development

(Extract from chapter)

7.4 Democracy and the Developmental State

As we have shown the requirements for the establishment of a developmental state and the emergence of democracy - initially an agrarian one largely coincide, the only divergence being that of the political rules of the game. We have also shown that building consensus on the rules of the game is not only consistent with the requirements of a developmental state but may also reinforce and consolidate it. But a number of apparent divergences have been pointed out by some analysts. We shall now address the more important ones.

One of the concerns has been with the "hustle of democracy". It is argued that the developmental state should be single-mindedly focused on doing what is needed to accelerate growth. If it also has to deal with democratic legitimization of its rule, not only will it be forced to spend a lot of time in doing so, but it may be forced to engage in patronage and socially wasteful rent-seeking activities. It could obviously be reasonably argued that democracy is so important that if this is the price to be paid for having it - so be it, a limited reduction in growth that may ensue is not too much of a price. However valid such an argument might be it assumes that the structure of all democratic politics is based on patronage and rent-seeking, and a fascinating study carried out in Italy has shown that this is not necessarily so.

Italy is as homogenous a country as it is possible to get when it comes to language ethnicity etc. But in many ways it is also two countries in one. The North and South have had such divergent historical and social evolution prior to unification that even after unification there has been a very clear divide between the two.

In the North there is a very dense network of civic organizations of all types and an individual is usually a member of a number of them at the same time. Such networks are horizontal and based on mutuality. People actively participate in public affairs. There is a large measure of trust. People assume and expect that the law will be obeyed by everyone. People including politicians are relatively honest. Politicians and the people value equity and tend to seek mutually beneficial solutions, they do not take politics as a zero-sum game.

In the South civic organizations are very thin on the ground and those that exist are mainly of the vertical linkage variety. The south is the home base of what has been called "amoral familialism" the family is the key "network" and amoral organizations such as the mafia have been based and superimposed on it. People are not involved in public affairs, they leave it to the "bosses". Political parties are important but involvement in them is based on their role as patronage machines. Corruption is the norm and it is taken to be the norm by the people and the politicians. There is very little trust and everyone expects everyone else to violate the law. As a result they seek someone else to enforce the law and impose it.

The North has ample social capital, the South has very little of it. The North has ample civic virtues, the South has very little of it. The North has been the home base of democratic and progressive thinking in Italian politics. The South has been the home base of the Mafia and anti-democratic tendencies in Italian politics. Since the introduction of regional governments in Italian politics, the North has had effective regional governments that people are satisfied with and politics in the regional arena has been relatively free from patronage and rent-seeking. The South has had very weak, ineffective and corrupt regional governments. Regional politics has been riddled, through and through with patronage and rent-seeking. If the South had been a separate country its politics would have been comparable only with Africa and not with Europe.

The path of economic development of the two parts has also been very divergent. About ten years after unification, in 1881 Italy was still a predominantly agricultural country. The South and the North did not differ much in terms of economic development. 60% of the population of the country lived in the rural areas, only 15% was engaged in industry including in cottage industry. The South was slightly more urbanized than the North. But Northern Agriculture was more productive and as a result per capita income was 15% - 20% higher in the North than in the South. By 1911 the North had made enormous strides in industrialization while the South had become less urban. Per capita income differences had increased to 50%. Despite the huge amounts of money thrown at the South to accelerate its development and the many twists and turns of Italian politics in the twentieth century, the South continued to lag and by the mid eighties, the difference in per capita income had reached 80%.

High social capital in the North has been blessed with virtuous circle, low social capital in the South has been cursed with a viscous circle, no policy turn, no amount of money has been able to overcome those differences and reverse the differing economic fortunes. The North has had a flourishing democracy, which is relatively free of rent-seeking and patronage. It is based on mutuality and public spiritedness. One can legitimately doubt as to whether the South would have been democratic if it had been a separate country, but as part of Italy it has had democracy but its politics has been patronage politics par excellence.

The experience in one and the same country tells us patronage and rent-seeking is not a necessary characteristic of democracy. It depends on the structure of politics. Where patronage is low, where social capital is high and public spiritedness adequate, democratic politics can be relatively free from patronage and rent-seeking. Where social capital is low, patronage high, public spiritedness low, all politics cannot but be riddled with patronage and rent-seeking. The political structure, which accelerates development is also the same structure that is relatively free from patronage and rent-seeking, and that which inhibits accelerated growth is the one which is riddled with patronage and rent-seeking and rent seeking.

Even if a developmental state was to be solely concerned about accelerating growth, it would have to build the high social capital that is vital for its endeavors. It would have to stamp out patronage and rent-seeking. These are the very same things that create the basis for democratic politics that is relatively free from patronage. A successful developmental state would thus be very well placed to be both developmental and democratic.

There is a catch, however. When a developmental state is established it is unlikely to find a situation where rent-seeking has been stamped out, social capital accumulated etc. If that were

the case the country would have been engaged in accelerated growth even before the establishment of a developmental state. It is therefore the developmental state that will have to prepare the ground and accelerate development at the same time. Initially therefore the risk of democratic politics becoming riddled with patronage and rent-seeking will be there. A more subtle argument has therefore been how can the developmental state clean-up the mess of patronage and rent-seeking in the initial states of its establishment by anything other than undemocratic means?

A related issue has been the need for continuity of policy. Developmental policy is unlikely to transform a poor country into a developed one within the time frame of the typical election cycle. There has to be continuity of policy if there is to be sustained and accelerated economic growth. In a democratic polity uncertainly about the continuity of policy is unavoidable. More damagingly for development, politicians will be unable to think beyond the next election etc. It is argued therefore that the developmental state will have to be undemocratic in order to stay in power long enough to carry out successful development.

Neither of these two related concerns could be dismissed off-hand. That is perhaps one of the reasons why democratic developmental states have been an even rarer species than developmental states in general. But those states that have played a developmental role and have done so in a democratic fashion, such as the social-democratic coalitions in some Scandinavian countries and the center-right coalition in Post Second World War Japan, the so called dominant partly democracies can point to one way out.

Studies have shown that stable long-term coalitions which stay in power for a long period but do so by democratic means can provide the needed continuity and stability of policy. The typical examples in these regard have been coalitions based on the labor movement and the middle classes in some Scandinavian countries, and coalitions between rural population and the right in Japan. The ruling coalitions in these countries have had regular, free, open and fair elections, and the basic political and human rights have been respected. They thus fully qualify as democratic regimes. But they have won elections repeatedly and have been in power for long-stretches. In the case of Japan the ruling coalition has been in power for almost 50 years.

A critical issue is therefore can such a stable, democratic and at the same time developmental coalition be established in a developing country. It is not difficult based on our analysis so far, to identify who the candidates of such a coalition can be. One group that cannot be part of the coalition is the private sector. One of the defining characteristics of a developmental state is that it must be autonomous from the private sector. It must have the ability and will to reward and punish the private sector actors depending on whether their activities are developmental or rent seeking. It cannot do so if the private sector is in the coalition. Obviously it does not mean that the coalition will have to be hostile to the private sector. It cannot be hostile to the private sector and bring about accelerated development in the context of the market economy. In the end, what the developmental state does will strengthen the value creating part of the private sector more than any other alternative. It only has to be independent from the private sector while at the same time doing things that will punish the rent-seeking part and reward the value creating part of it.

Any democratic state, developmental or otherwise, in a developing country will have to be agrarian at least in its initial phases. The other alternative is to wait until a substantial business and middle

class has been created i.e. until much of the work of accelerated growth has been carried out. The problem, however, is that without a developmental state, most if not all of these countries will be stuck in the poverty trap and the substantial business and middle class will not be created. The question therefore is whether a developmental state can be firmly based in the rural areas and whether it can use this to establish a stable coalition that rules democratically.

We have shown that agriculture is and must be the engine of accelerated growth at least in the initial period of the process of development. We have argued that widespread and relatively equitable ownership of assets is a requirement for accelerated development of agriculture. It is clear accelerated agricultural development will have to include commercialization, it cannot be based on sustaining subsistence farming. We have also argued that various local voluntary organizations to support marketing and improvement in productivity have to be established. Finally we have shown that resource transfers from agriculture must be such as to maintain the incentives for farmers to continue to increase production.

All of the above are fully consistent with the interests of the farmers. One can even claim, that it is very difficult to envisage any other package that would be more in tune with the interests of the farmers. The rural population can therefore be the solid base for a stable developmental coalition in a developing country. The steps that have to be taken to accelerate agricultural development are also the steps that are needed to bring about the changes in the social structure of the peasant to transform him/her into a force for democracy. The activities of a developmental state will thus not only be consistent with the interests of the peasants but also with their social transformation into a force of democracy. There will be the normal limitations of a dispersed rural population, but we have seen that under certain circumstances it does not become an insurmountable hurdle for democracy.

The only other fly in the ointment has to do with resource flows from agriculture. This could be a source of tension between the developmental state and the peasants. But in any case every developmental state will face the problem and will have to resolve it in a manner that maintains the incentive for the farmer to produce more and improve his/her income in so doing. Potentially, therefore, the peasant is the bedrock of a stable developmental coalition. With the votes of the peasants who constitute the bulk of the coalition, with the democratic potentials of a socially transformed peasant, the developmental coalition will have what it needs to rule democratically to ensure continuity by democratic means and to stamp out patronage and rent-seeking activities. A coalition based on the very sector, which has historically been the victim, rather than the beneficiary of patronage and rent-seeking activities will have all the will to stump it out.

The urban middle classes and labour, however small they might be could also be members of such coalition. It has of course been argued that they do benefit from patronage and rent seeking in the context of what has been called urban bias. The crumbs they may get from such activity is however nothing when compared with what they could get from job creation in the context of accelerated growth that is broadly equitable. They tend to be much easier to organize than the rural population and under the appropriate environment more democratic in their orientation than even the socially transformed peasant.

A coalition that covers much of the rural and urban population but is firmly based on the rural base, that includes all those that have very little to gain from patronage and rent-seeking, a coalition that

includes the vast majority of the population and hence can guarantee continuity through the democratic process would be a solid base for a state that is both democratic and developmental. Such a state would in effect be one form of the so-called dominant party or dominant coalition democracy. Such a state based on a solid and dominant coalition of forces who have no stake in patronage and rent-seeking would be able to avoid and overcome socially wasteful patronage and rent-seeking.

Technically policy stability and continuity could be achieved even when parties regularly replace each other in governing the country. But this can be so only where such a solid consensus among politicians and the population on fundamental policy has been achieved and where politics is confined to dealing with trivialities and personalities. Such a situation is very unlikely to emerge in a developing country. In addition politics based on personalities can easily degenerate to patronage politics. The most likely scenario for a state that is both democratic and developmental to emerge is in the form of a dominant party or dominant coalition democracy.

Most of what a developmental state has to do in order to be a developmental state are also the things that need to be done for a stable democracy to emerge in a poor developing country. The critical additional step required is to establish a solid developmental coalition to govern the country democratically. The basis for doing that are those steps that a developmental state will have to take any way. There is therefore no reason why a developmental state should necessarily become undemocratic. There is every reason to suggest that if a developmental state were to also be democratic the "hegemonic" nature of its development project would be achieved faster and held more deeply because it would emerge from free debate and dialogue. A democratic developmental state is thus likely to be even more effective as a developmental state than an undemocratic one.

In previous parts of this chapter we have shown that a stable democracy can emerge in a poor country and what the requirements are for such a polity to emerge. They largely coincide with the requirements for the emergence of a developmental state. Where the circumstances for the emergence of a developmental state do not exist, the circumstances for the emergence of a stable democracy in a poor country do not exist. One can therefore conclude that the prospects of a stable democracy in a poor country are intimately related to the establishment of a developmental state and achieving accelerated development. In poor developing countries, a developmental state, accelerated development and stable democracies appear to be parts of the same package.

The only exception one can make is that accelerated development and developmental state can occur in a non-democratic polity. But that would not change the basic conclusion. Where the circumstances for a developmental state do not exist the chances for a stable democracy to emerge are indeed very remote. Where they exist while there is no guarantee for democracy, there is a reasonable chance for a developmental and democratic state to emerge. In the end, therefore, the chances of a stable democracy in a poor country are related intimately to the emergence of a developmental state and accelerated development associated with it.

8. Background to The Miracle

(Extract from chapter)

8.5 Conclusions

Japanese colonialism came to an abrupt end at the end of the Second World War in 1945. Taiwan was immediately returned to China, and the Kuomintang which was in power in mainland China ruled Taiwan from China up to 1949. When the Kuomintang was defeated, it fled the mainland and established itself in Taiwan, initially claiming to represent the whole of China. Since 1949 Taiwan has been ruled as a de facto independent country. Korea was divided into two parts; the north under the control of the Russians, the south under the control of the Americans. An American military administration ran South Korea up to1948, when it was replaced by a South Korean government. Soon after that the Korean war started and devastated the region. After the end of the war, attempts to rehabilitate the economy were made with limited success. The political turmoil, which resulted from such limited success led to a military coup in 1961, and it is under the military regime that the Korean miracle took place.

There was thus a transition period between the end of Japanese colonialism and the beginnings of the economic miracles of the two countries. Japanese colonialism was deeply repressive and a traumatic experience for the peoples concerned. Its objectives had nothing to do with the interests of the peoples of the colonies. It was thus as detestable as any other colonial regime. Japanese colonialism was no better or no worse than European colonialism. Attempts to paint Japanese colonialism as somewhat better, as something guided by "co-prosperity" of the Asian people etc. is just mere colonial propaganda. Nevertheless Japanese colonialism had its own characteristics that had significant impact on the future development of the two colonies.

Japanese colonialism was established by a developmental state, which had not yet completed the process of industrialization in its own country. Naturally the colonial regime reflected the practices and experiences of the home government. The Japanese colonial state was just as interventionist as the developmental state in Japan and its activities in agricultural development were patterned along the lines of Japanese agricultural development. Later on when industrialization began in the colonies it was state led industrialization with the involvement of Japanese companies as subordinate participants in the exercise. All the instruments of the developmental state in Japan were adapted and used in the colonies. The experience of the two colonies with the development of market economies was thus one of development engineered by a developmental state that made effective use of all the instruments of such a state. The two colonies had an early and successful introduction to the nature, instruments and characteristics of a developmental state.

Japanese colonialism focused as it was on extraction of an agricultural surplus and supplying the home market with rice and to some extent sugar could not but aim to transform the rural economies of the colonies as a whole. It could not be based on enclave large-scale farms and on a hands-off approach to much of the rural areas either. It thus deeply penetrated the rural areas and established a very powerful, effective and repressive state apparatus down to every village of

the colonies. It paved the way for the creation of a state that is both effective and towers like a colossus on society.

Japanese colonialism built the infrastructure, and spread the primary education that is essential for agricultural development. Both the education and the rural infrastructure covered all the areas of the country and involved the bulk of the population. The peasants were also introduced to new technologies and commercialized farming. However all of these were done in a harshly exploitative fashion based on the dominance of landlords, who were themselves being commercialized. The impact on sustained and accelerated agricultural growth was thus limited. However both the infrastructural, and technological basis for later growth were very well prepared by Japanese colonialism. However limited industrialization may have been under the Japanese, it was the basis on which further industrialization could be contemplated.

9. Agricultural Development in Taiwan and Korea

(Extract from chapter)

9.6 Conclusions

Land reform and agricultural development in Taiwan was exceptionally successful. The Land reform removed rent-seeking from the rural scene, stabilized the regimes politically, paved the way for equitable development and facilitated resource extraction from the farmers by the state. The development of cooperatives played a crucial role in removing market failures in both the credit market and in marketing of agricultural products. Universal primary education, effective research and extension work and government leadership in developing and implementing a trajectory of land augmenting technological development addressed the key market failure in agricultural technological capability accumulation.

The speed and nature of agricultural development in Taiwan was such that its multiplier effect on non-agriculture was maximized, and established a firm basis for the flourishing of small enterprises which became the basis for export led industrialization. There was no liberalization of agricultural prices but the prices that farmers got in Taiwan did not inhibit the farmers' incentives to produce more. As a result national savings and economic growth were very high.

While land reform and agricultural growth were also successful in Korea, they were not as successful as in Taiwan. The size of farm land and the climate in Korea were less conducive to agricultural growth. More importantly, the government did not devote enough attention to agricultural growth until the 1961 coup and agricultural prices were kept too law until then. As a result agricultural growth started late, and Korea had to launch its industrialization drive without significant national savings and without a significant non-agricultural small enterprise sector. This had a big impact on the structure of the Korean economy.

The experience of the two countries shows what type of deep and massive market failures impede agricultural development and how an effective state intervention can overcome these problems. It also shows that fast and equitable growth in agriculture is the start of a successful and equitable development.

10. Overcoming Market Failures

10.4 Conclusions

The governments of Taiwan and Korea, in effect, largely replaced the financial market and allocated investible resources in accordance with their development plan. Credit allocation by government was the linchpin of a comprehensive package of support to the private sector. The two governments also provided a minutely differentiated trade support and protection to support infant industries. They used various instruments to coordinate private sector investment. In other words the governments of the two star performers intervened massively to address all the key market failures, and used a comprehensive set of instruments -- instruments that were not only consistent with each other but supplemented each other.

The two governments were unique not only in the extent and mode of their intervention in the economy but also in ensuring that the private sector faced not just support and reward but also sanctions and penalties. Only those who delivered in terms of globally competitive goods and services benefited from the comprehensive support that the governments provided. Massive, and comprehensive government intervention to address market failures combined with the appropriate set of incentives and disincentives provided to the private sector to elicit the desired response was how the Taiwanese and Korean miracles were achieved.

11. Accumulating Technological Capability

(Extract from chapter)

11.6 Conclusions

Taiwan and Korea are without doubt the most successful of all developing countries in terms of achieving successful development over the past 50 years. They have had the highest sustained growth rates (with the exception of China over the past few decades). They have had the most equitable growth of all developing countries. They have had the most rapid and sustained accumulation of technological capability and growth in TFP. Taiwan's performance has been superior to that of Korea on all three counts, but Korea's performance has also been spectacular by any standard.

These most successful of development experiences have not been brought about by a night watchman state restricted to protecting individual and property rights and enforcing contracts. It has been brought about by some of the most interventionist governments to emerge in the context of a market economy. As we have seen in previous chapters most if not all of the prescriptions of the neo-liberal paradigm have been massively and systematically violated by the two regimes. The practices of the two most successful development experiences over the past half century are in direct and total opposition to the neo-liberal paradigm.

The activism of the two governments has, however, not been haphazard or wanton. They have intervened in precisely the areas where theory suggests there are pervasive market failures. The instruments used to do so by the two governments have been similar but by no means identical. The two countries have selected instruments that are consistent with their differing economic structure, and while Taiwan has been more successful, the instruments used by Korea have also led to success. Massive and systematic government intervention in the economy has not resulted in massive rent-seeking and economic failure as the neo-liberal paradigm would suggest but in spectacular success. The experience of Taiwan and Korea shows that the neo-liberal paradigm is unable to account for the most successful development experience in the post war period. Only an explanation based on the need for activist governments to address market failures can adequately account for the success of Taiwan and Korea.

12. The Political Economy of Development in Taiwan and Korea

(Extract from chapter)

12.6 Conclusion

The establishment of developmental states in Taiwan and Korea are the results of a complex combination of factors. The right social environment for a strong state firmly established in the rural areas was created by the land reform which was itself the result of complex national and geopolitical factors. The single minded determination to pursue shared growth and accelerated development was the result of the challenge of communism that the two regimes faced and the determination of the leadership to do what ever it takes to overcome the challenge.

Once a political leadership which felt its survival and legitimacy depended on shared growth took power in an environment where land reform had created a conservative mass of small farmers, the rest could be done by the leadership itself. The leadership was able to change the playing field and the rules of the economic game in favor of productive activities and developmentalism. Efforts at consensus building reinforced by the success in achieving shared growth could be made with a high probability of success. The bureaucracy could be remodeled to fit the development agenda.

As a result of favorable national and geopolitical circumstances and of effective and well considered action on the part of a committed leadership, two states that were predatory were transformed into developmental states. To that extent we can say that developmental states are created not inherited, we can say that they are results of political action not handed down by providence.

13. Dependency and Development in Taiwan And Korea

(Extract from Chapter)

13.4 Conclusions

We have in the previous chapter explained the success of Taiwan and Korea purely in terms of domestic economic and political processes. But obviously that cannot be the whole story. External developments did and could not but play a vital role in the development process of the two countries. In particular, the U.S. provided a comprehensive package of support including security guarantees and non-reciprocal market access to the two countries. Some of these supportive measures such as providing non-reciprocal market access were exceptional indeed because the U.S. government had to withstand pressures from rent-seeking constituencies in the U.S. to provide such support. It was the exigencies of the cold war, which enabled the U.S government to do so. Nevertheless the fact remains that the decisive causes of success in Taiwan and Korea are domestic factors, particularly the political factor of the establishment of developmental states in the two countries.

Various schools of thought have tried to "explain" the success of Taiwan and Korea. The dependency school of thought suggests that the two countries are not really developed but semi-peripheries, and that they were allowed to reach such a stage by way of exception. This school of thought thus suggests that even the "semi-peripheral" states of the two countries is exceptional and one that can not be repeated by other developing countries.

The neo-liberal school of thought suggests that what really precipitated the accelerated development of the two countries was the policy of letting the market do its wonder. Indeed, as it has become very clear that the two governments intervened in the economy massively, some in the neo-liberal school have gone as far as to suggest that much of government intervention nullified each other and simulated a free market.

The dependency school is in conflict with the success of development not only in Taiwan and Korea but also elsewhere in the developing world. It is a theory of underdevelopment unable to explain development in developing countries in the context of a market economy. The neo-liberal school is in conflict with the facts of development in the two countries and with much of historical experience of sustained, accelerated growth.

The school of though which has at times been called the revisionist school based on the analysis of market failures and developmental states is capable of explaining the successful development of Taiwan and Korea successfully. It is also consistent with the facts of the historical process of development in many of the developed countries

14. The Genesis and Crisis of the Predatory State in Africa

(Extract from chapter)

14.4 Conclusions

A number of fundamental conclusions about the genesis and crisis of the predatory state in Africa can be made on the basis of the assessment made in this chapter

Colonialism in Africa was primarily extractive rather than productive. It created a weak and predatory colonial state. It established a rent-seeking political economy based on enclave production, production of tropical agricultural products for export, and made the state the primary instrument of wealth accumulation. The national exchange systems and systems of authority were very weak and were buttressed primarily by networks of patronage. Africa was not a significant destination for foreign capital during or after colonialism. No institutional and physical infrastructure for the sustained development of Africa was built during colonialism or after it, largely due to the marginal role of Africa in the global system. The marginality of Africa was thus a key source of its problems.

The key problems of Africa emerged not because Africa was an important destination of foreign capital, not because Africa's economy played an important role in the global economic system but because Africa was marginal as a destination of foreign capital and marginal in the global economic system. It is because of such marginality that colonialism in Africa did not even try to build the physical and institutional basis for vigorous capitalist development. It is because of such marginality that the limited capital that was attracted to Africa was largely of the extractive and rent-seeking variety. It is because of such marginality that African diplomacy was not based on trade and investment but on diplomatic clientage. It is such marginality which was the basis of the dominance of rent-seeking activity in every aspect of its external relations. It is such rent-seeking which is at the root of African problems.

The African state which emerged from colonialism had essentially the same characteristics irrespective of the socialist ideological pretensions of a significant number of them. It was a predatory state driven by the logic of maximizing rent for the rulers whatever its implications for the economy. Its capability to collect rent was based on its intermediary status. It was the intermediary between its people and the international system, and collected rents through foreign aid in return for diplomatic clientage, through providing privileged access to domestic resources and market and taxing trade.

The predatory African state was able to sustain the expanded demand for rent-seeking and patronage in the first decade of independence. But the dynamics of the economy were such thatthe growth potential became exhausted quickly and the sources of rent on which the whole system depended began to dry. External shocks accelerated the process of decay. The response of society to the crisis was not one of fundamentally transforming the rent-seeking political economy but one of exit. The response of the predatory state to the crisis was not to reform the system but to continue the process of rent-seeking at the risk of speeding up the implosion of the whole system. The predatory state and the rent-seeking society were thus heading for a

catastrophic implosion and were in an advanced stage of dissolution when the international system began its program of salvaging the situation by transforming Africa along the lines of the neo-liberal paradigm.

15. Botswana: The Exception That Proves the Rule

16. The Neo-liberal Diagnosis and prescription

17. Outcome of Economic Reform

17.1 Agriculture Under Economic Reform

The centerpiece of agricultural reform in Africa has been, the deregulation and liberalization of markets, getting the price signals right and using price signals to kick-start sustained agricultural and economic growth. Conceptually the reform did not preclude investments in infrastructure, research and education of farmers. Credit markets were to be largely taken care of by the informal sector. The empirical foundations of the significance of getting the price signals right, however, appear to be very shaky indeed.

Econometric studies have estimated that the long-term price elasticity of supply for agricultural crops in Africa to be approximately 0.3 i.e. for 10% rise in price, supply will increase by around 3%. The World Bank's own studies suggest that only 10% of the variations in agricultural growth appear to be related to price variations. Response of trade volumes to transportation cost has however been very high. World Bank studies suggest that a 10% drop in transport costs increases trade volumes more than two times i.e. by 25%. Transportation cost of traders in agricultural products are very high. Studies have shown that such costs in Africa can be as high as 60% of the operating costs of the traders.

Empirical analysis thus shows that the long-term supply response to prices of agriculture is low, that as much as 90% of variations in agricultural production cannot be explained by price variations and thus that price variation while important is not the main explanatory variable of growth in African agriculture. The neo-liberal paradigm's insistence on price signals as the key that unlocks sustained agricultural growth has thus not only no firm theoretical foundation as we have shown in part II but also no firm empirical foundation as the figures above indicate.

And to the extent that prices play a role transportation costs appear to play the key role. Every percentage reduction in transportation cost appears to increase trade volumes and hence the main item in the volume of trade in Africa, trade in agricultural products. As transportation cost can be as high as 60% of traders operating costs reducing the transportation cost by half would nearly double trade.

The Bank reform program linked getting price signals right with liberalizing and deregulating marketing. In almost all reforming countries market entry by private traders has occurred in food and cash crop markets as well as in fertilizer markets. While entry has been extensive at the small-scale level private sector entry into the whole sale trade, motorized transport and external trade have been limited to those groups with "strong social networks and state connections." Thus private sector involvement has been quite extensive at the bottom end but has in effect been cornered at the top end by few traders with "strong social networks and state connections". It has at best become an oligopoly at the top end fed by a wide network of small and petty traders at the bottom.

Traders have had limited access to credit, and face high transaction costs related to, among other things, obtaining market information, searching for buyers or sellers and enforcing contracts, poor communications infrastructure, lack of an effective system of grades and standards etc. As a result grain markets remain risky, personalized with limited long-term investment in transport or storage, and highly volatile cereal grain prices.

Agricultural marketing after liberalization continues to be very inefficient. The state is not providing a system of grades and standards and a means of effective contract enforcement that even a night watchman state is supposed to provide. There is no effective system for the collection and dissemination of market information. The private sector involved in agricultural marketing is weak and dispersed at the bottom end and oligopolistic at the top. There is very limited investment in grain marketing structures, markets are "personalized and volatile". Liberalization has not unleashed the market. Creating an efficient agricultural marketing system is not merely a matter of "removing the dead hand of the state". That is why liberalization in Africa has replaced one inefficient state-led system, with another inefficient system led by well connected oligopolists.

The combination of high transport costs, fragmented, and oligopolistic private sector marketing at two ends of the market, high transaction costs related to institutional inadequacies have meant that FOB prices of African agricultural exports are often two to three times higher than farm gate prices whereas farm gate prices of inputs are two to three times higher than their CIF price. The agricultural marketing system and the very high transaction costs involved in it in post reform Africa is such that despite the liberalization measures African agriculture can at best be described as semi-open. This cannot but have a huge impact on the prospects of agricultural growth in Africa.

The lending and saving cycle of the informal financial sector in Africa and its localized nature limits the role of the sector in term financing for agricultural investment. Much of the limited credit available at high interests is thus mainly for smoothing consumption rather than for investment. The potential use of savings mobilized through the informal sector for economic growth and diversification "remains unrealized". The result of leaving agricultural credit to the informal sector so that it may provide credit "at lower social cost" has not only been the traditionally high interest payments but the lack of availability of adequate credit for agricultural investment. Its result has been to starve small scale agriculture of credit.

The neo-liberal paradigm allows for government expenditure in research, infrastructure and farmer education. But as we have shown in the previous chapter capital spending has been lower than the already low pre-crisis spending and this could not but affect infrastructural investment. Indeed

since the 70s when the crisis began infrastructural investment has been neglected. As a result Africa which has been at the bottom in terms of quantity and quality of infrastructure has fallen behind even further over the past 15 years. Investment in Agricultural research in Africa has actually declined from \$701 million in 1981 in the pre-reform period to \$684 million in 1991 during the reform period.

The reform program allows for investments in agricultural research, education and rural infrastructure, but actual investments in these areas in the reform period have been lower than the already low investments of the pre-reform period. This is so because the other elements of the neo-liberal reforms, particularly reforms in the fiscal and financial area make it impossible to allocate adequate funds for investment in those areas. This is so also because the predatory state in Africa had never had the commitment to invest in the physical and institutional infrastructure required to transform agriculture. Like the colonial state it inherited the African state's interest in the rural areas was limited to maximizing the collection of rent.

The semi-open nature of African agriculture of the reform era, which is caused by the inefficiency of the liberalized agricultural market leads to a vicious circle. The extremely high marketing costs and low productivity means that much of agricultural production is non-tradeable. About three quarters of rural non-farm income and employment is related to services of all kind, handcrafts, fresh meat, milk and other perishable products which are also largely non-tradeable. Thus both agricultural and rural non-farm activities are largely non-tradeable, their production and consumption will depend on local markets and hence local income. In such an environment the initial low income of agricultural production constrains and limits the growth of rural farm and non-farm income. Rural Africa in the reform period is demand constrained. Because it is demand constrained a significant share of rural primary resources can remain underused for long periods of time, even if macroeconomic and trade reform remove price distortions.

Agricultural growth can thus remain within confined limits more or less indefinitely unless the demand constraint is removed. For the demand constraint to be removed agricultural products must be sold outside the locality in greater volume and /or value. This would increase farm income in the locality which would in turn increase the demand for non-farm activities. Increased non-farm income would in turn increase demand for agricultural products within the locality and so on in a sort of a virtuous circle. The key that unlocks this virtuous circle -greater sale outside the locality -materializes only if transaction costs come down dramatically and productivity increases in a similar fashion. A combination of technological and productivity change and dramatic reduction in transaction costs are critical to break the logjam that has narrowly circumscribed African agricultural growth.

But as we have shown liberalization of agricultural markets has not reduced transaction costs. Rural infrastructure is lacking. Marketing infrastructure such as storage facilities and transport are lacking. Rural credit is lacking. At the bottom end there are many small and weak operators unable to provide the economies of scale, at the top end well-connected traders have monopolized trade. Market support institutions such as market information systems and standards and grades systems are absent and contract enforcement is weak. Neither the state nor the private sector has delivered in terms of bringing down transaction costs to reasonable levels.

Productivity increments and technological change are inhibited by continued reduction and

stagnation in the minuscule investments in research and extension of the pre-reform period. Investments in infrastructure such as irrigation which have a big impact on productivity and technological change have been and continue to be insignificant. The institutions and investments that are required to bring about technological change in agriculture have been remarkably absent or weak beginning with the colonial period upto the pre-reform and post reform periods of independence.

While African agriculture has thus been plagued by high transaction costs and near absent productivity growth, South East Asian countries which produce similar agricultural products for export have been increasing their productivity. Their productivity is as much as three times higher than that of Africa. Africa is thus being squeezed by high productivity competitors.

The neo-liberal paradigm may have removed the dead hand of the state, which had stifled agricultural growth but it has replaced it with the deaf ear of a liberalized market with pervasive market failure. The continued decline in prices of Africa's agricultural exports, the low productivity of African agriculture, and competition from more productive competitors is sooner or later going to squeeze Africa's exports out of the world market. In practice the predatory states "statist" rent-seeking policies, and the neo-liberal paradigm's mantra of unleashing the market have miserably failed in unleashing African agricultural growth.

It is not difficult to see that this trend cannot continue without precipitating a catastrophe in Africa. The continued decline in world markets of Africa's agricultural exports, the low productivity of African agriculture and competition from more productive competitors—is sooner or later going to price Africa's exports out of the world market. Unless Africa's agricultural technological capability increases, population growth is likely to create a Malthusian reckoning. Africa's Agriculture has reached a dead end.

Investments in infrastructure and market support institutions that lower the unit costs of distribution are needed to break the dead end. The physical and institutional infrastructure of technological capability accumulation must be built to bring about continued improvement in productivity and break the dead end.17 In other words the dead end can be broken only if government invests in rural physical infrastructure, in market support institutions, in institutions for technological capability accumulation, and if government intervenes to save agriculture from the credit crunch that the informal sector has subjected it to. The neo-liberal dogma is opposed to all of the key steps required to break the dead end. The neo-liberal paradigm is thus fundamentally incapable of leading African agriculture out of the dead end that it has reached. The fundamentals of the neo-liberal paradigm go against the grain of what is needed to bring about rapid and sustained growth in African agriculture.

17.2 Finance and Economic Reform

Banks in the reform period have exhibited a number of common characteristics across Africa. They have, however, been dominated by two characteristics in particular: an extremely high incidence of non-performing loans and excess liquidity. Various explanations have been given for these characteristics. One such explanation is that Banks in Africa lack "information capital".

Indeed in Africa information about potential borrowers and in some instances about the economy in

general is lacking. Tax returns are usually fraudulent and unavailable to third parties. Audited reports are rarely required and when they are available are highly unreliable. The income and property of citizens is not properly registered and tracked. There are very few if any institutions that collect and analyze information about firms and businesses. Thus, when banks allocate credit they are operating almost in the dark. In such circumstances they are likely to give credit to those who are not really credit worthy and hence the high incidence of non-performing loans. The lack of adequate information and the associated high incidence of non-performing loans is also likely to make them excessively risk averse, and hence the excess liquidity. Indeed lack of adequate information is one of the problems that plague the financial markets even in better information endowed countries. In Africa with almost a total lack of adequate information, it would be expected that it would be one of the key bottlenecks for the efficient operation of banks.

It has also been argued that the high incidence of non-performing loans and excess liquidity is due to lack of demand for credit from profitable private sector borrowers. This in turn is blamed on poor supply response to the reforms and the reduction of the role of large firms in the economy and the proliferation of small and informal enterprises.

Indeed the fact remains that growth of African economies has been low and erratic. The supply response to the reform has been low. Such an environment would not create too many opportunities for profitable investment and hence the lack of demand for credit in capital scarce countries of Africa. The lack of demand for credit would explain the excess liquidity and the weak supply response could also be associated with the financial problems of borrowers and their inability to properly service their loans and hence the high incidence of non-performing loans. In the reform period, the role of larger enterprises in the economy has diminished whereas as the role of informal and small enterprises has increased. Banks find it easier to lend to large corporate customers because the risks and more importantly the information failures are lower among the large corporate customers. The small and informal sector is however effectively "unbankable", from the point of view of the banks because of the risks and information failure. The reduction in the bankable sector and the increase of the "unbankable" sector would be one explanation for the excess liquidity.

Other explanations relate the high incidence of non-performing loans and excess liquidity to another characteristic of post reform African banks: high interest rate spreads and very high rates of real interest rate, which have been the results of reforms across the continent.

Very high real rates of interest generated by financial liberalization depress investment. Indeed the World Bank's study of the East-Asian experience has shown that financial repression can be a powerful instrument to promote growth and investment. One of the objectives of the reform program in Africa was to undo financial repression and it has succeeded admirably. The consequence would naturally be to depress investment, reduce demand for credit and generate excess liquidity. High interest rates are also known to increase moral hazard and default risk.

The financial sector was plagued with rent-seeking activities before the reform period. Indeed one of the justifications of the reform was precisely to reduce such activities by privatizing government banks and eliminating the role of government in credit allocation. Banks have been privatized and the role of government in credit allocation has been eliminated. But rent-seeking in the financial sector has not been overcome as the experience of several countries amply demonstrates.

Financial liberalization can open up new rent-seeking opportunities in four different ways: (a) substantial premiums were obtained by getting foreign exchange at auction prices and earning a margin on resale to end users or other market operators, (b) financing of foreign exchange purchases by providing loans at retail or interbank rates, (c) arbitrage in money markets, taking advantage of interest rate differentials, (d) pyramid schemes and capital export i.e. plundering the financial system through blatant fraud.

Such large rent-seeking opportunities attracted capital and manpower to the financial sector. Its contribution to GDP increased almost four fold in the wake of liberalization, and by 1992 its share of GDP approached 9% exceeding that of manufacturing, which had declined since the mid 1980s. This fantastic growth in the financial sector was taking place at a time of declining investment, which had declined from 19.4% of GDP in the pre-reform period of 1986 to 15.6% of GDP in 1992 after the reforms. Savings had also declined from 31% in 1985 to 19% in 1993. At a time when the number of banks was tripling and the share of the financial sector in GDP was quadrupling in the wake of financial sector liberalization, lending to the private sector was actually decreasing in real terms. Lending levels to the private sector in real terms in 1993 were 60% of what they were in 1986-87.

Excess liquidity that has plagued banks during the reform period and high incidence of non-performing loans are thus caused by a number of factors. The massive information failure in Africa, the weak supply response and informalization of the economy, the high interest rate spreads and high real interest rates which have priced out much of the potential investment in the economy and increased default risk, and pervasive rent-seeking are the main causes of the phenomena.

Deposit mobilization by commercial banks has also been affected. Excess liquidity means that banks do not seek additional savings when the deposits they already have are sitting in their vaults. Indeed interest bearing long-term deposits would increase their cost of fund at a time when lending is very low. Banks have therefore not only failed to be active in mobilizing savings but have also been known to at times turn savers away. What has emerged in the wake of financial sector liberalization is therefore not just excess liquidity and low levels of lending but also low levels of savings with one feeding the other .

The liberalization of the financial sector has increased the entry of foreign banks somewhat and increased competition. The foreign banks have had to focus on the most profitable segments of the market and these happen to be the largest urban centers and the bigger corporate customers. They have therefore had a trend of reducing their presence in the smaller towns and rural areas and reducing their service to customers outside the large corporate sector. The new entrants have started with such a narrowly focused approach while the older ones have had to shift towards such a narrow focus. Clearly then the presence of foreign banks has not affected the nature or scope of the problems of the post reform African financial sector. There has been little improvement in quality of service or range of service.

The neo-liberal Paradigm argued that it will revive the financial sector which had been run-down by the predatory African state through financial sector liberalization. It argued that it will resolve the problem by removing the dead-hand of the state from the financial sector. The results speak for

themselves. The continent is trapped in a low lending-low savings trap. Interest rate spreads and real interest rates have become very high effectively pricing productive investment out. There is excess liquidity in countries which have extreme scarcity of capital. Banks have no interest in actively mobilizing savings and are at times turning savers away in countries where savings are very low any way. The financial sector has completely failed in its task of financial intermediation. The neo-liberal paradigm is unable to recognize the fundamental market failures in the sector that can only be overcome through effective government intervention. It is unable to overcome its dogma of leaving the markets alone even when they don't work well. It is unable to lead Africa's financial sector out of the dead end that it has reached during the reform period.

The reforms have not only failed in bringing about efficient financial intermediation and have reached a dead end in this regard but have also failed to overcome rent-seeking in the financial market. The main actor in rent-seeking in the sector is no longer the state as state. The main actor is the private sector, although of course in Africa with so much "straddling" it is not always easy to identify where the one begins and the other ends. By narrowly focusing on rent-seeking by the state it has in effect opened widely the door for private sector rent-seeking. By limiting the effort of reducing rent-seeking activity to reducing the role of government in the economy, it has left the other aspect of rent-seeking, the private sector aspect almost intact.

Of all the variants of the neo-classical school of thought, the Washington consensus was perhaps better placed to understand the pervasive market failures in the financial sector, particularly in developing countries and even more particularly in Africa. Its analysis of East Asian experience could not have been clearer in recognizing these failures. But it came to the conclusion that government failure was worse than market failure, and that markets however imperfect will in the end muddle through. Letting markets be may not produce spectacular results when they are imperfect, but they will at least bring about adequate growth, it was argued. Practice has shown that market failures are as bad as government failures and both have led to a dead end. The predatory state and its "statist" policies, and the Washington consensus have in their own different ways ended up by leading the financial sector in Africa into dead ends. One dead end is as bad as the other. It appears both approaches have to be scrapped if the financial sector in Africa is to get out of these dead-ends.

17.3 Industrial and Economic Growth in the Reform Era

All the infrastructures, institutions and instruments for accelerated manufacturing growth in particular and economic growth in general are either very weak or absent altogether. This must be a critical factor in the dismal performance of the economy in general and manufacturing in particular.

Such an environment affects both the domestic private sector and FDI. But before we assess the impact of the post-reform period on FDI, we shall briefly review the results of the reforms on manufacturing and overall growth of African economies by contrasting the decade and half before reforms to a decade and half of reforms using different indicators.

The pattern is as clear as it is depressing. All the indicators used have been going down since before the reform period reached their crisis point, and have continued to decline despite some improvement since the 80s, which however, have not been enough to reverse the trend or reach

the levels of the early years of independence. There is one exception to this rule and that is manufacturing output. Its growth rate declined for the thirty years from 1965 to 1994, without interruption either in the 15 years before or 15 years after the reforms, and its performance during the reform period has been even worse than that during the early crisis period. Clearly then the figures show that manufacturing which has never done well in Africa, has done even more badly in the reform years. African industrialization has aged pre-maturely and has come to a dead end.

FDI was expected to change the picture somewhat, and despite the neo-liberal schools eloquent, not to say poetic, description of the informal sector as the saving grace of the otherwise dismal economic landscape of Africa, in practical terms, the effort of the reform program has focused an attracting FDI, and its performance reflects the depth of the problems of industrialization in Africa. Net private capital flows globally increased from \$43.5 billion in 1990 to \$226 billion in 2000. There has thus been a massive increase in private capital flows overall. Similarly FDI in Africa has increased by roughly 4 billion from \$1,683 million in 1991 to \$5,454 million in 2000. If we divide African countries into two: those who export minerals and oil and those who do not, the latter representing 71% of sub-Sahara Africa's population and 82% of GDP, we can see where the focus of FDI in Africa has been. The average for the oil and mineral exporters for 1991-1995 was \$1,063 million, and for 1996-2000 \$3,354 million. Whereas for the rest, the average for 1991-1995 was \$203 million and for 1996-2000, \$2,101 million. Clearly FDI has increased particularly since the mid 1996 in both groups of countries, but by 2000 countries with mineral resources and representing 29% of the population of SSA, were getting over 60% of the total FDI, whereas those who did not produce mineral were getting \$5.20 per capita in FDI.

The \$ 5.20 per capita FDI includes investment in privatized public enterprises and as there was a lot of such privatization in the 90s, new investment must have been significantly lower than that. Perhaps more importantly, much of the remaining investment in the non-mineral producing countries was also devoted to natural resource extraction.

In a study on natural resource development in Africa, David Kaimowtiz explains why. In a world of very high capital mobility, capital has the option of comparing returns across the globe and selecting the places that maximize returns. Those that attract capital are those that are superior to almost every other destination in this regard, or to use Kaimowtiz words FDI goes not to those who have "comparative advantage" but to those who have "absolute advantage". Unless a country has sufficient infrastructure, an educated and healthy work force, world-class managers and professionals and well functioning support institutions, it is unlikely to attract much FDI. The only thing to attract investment in countries that lack these conditions is not cheap labor per se but the opportunity to rapidly exploit a small set of valuable natural resources, such as minerals, timber, prime agricultural land and exotic wild-life that can not be obtained elsewhere at low prices. Thus the absolute advantage of even those who do not have mineral resources is in the natural resource area, and the FDI they attract is focused on its rapid and profitable exploitation.

The reforms do not address private sector rent-seeking and the development of natural resource through FDI has involved a lot of such rent. Indeed as David Reed says in the same study the reform programs have been designed to attract foreign capital by providing exceptional opportunities to obtain monopoly rents. It is through such rents that the FDI has been attracted to much of Africa. The more things change the more they remain the same. The things that attracted such personalities as Cecil Rhodes appear to be the things that attract FDI in the current very

different environment. Such FDI is of course practically irrelevant as far as sustained and rapid development is concerned. Thus even the truly authentic foreign private sector has continued to shy away from Africa and to engage in collecting rents when it gets engaged with it. Private rent-seeking in investment particularly FDI has not been significantly affected by the reform programs.

To say that private sector rent-seeking has continued unabated during the reforms is however not to absolve state actors from the practice. Indeed where there is rent-seeking, corrupt state officials will not be far off. The same study on natural resource development in Southern Africa states that corruption and collusion have accompanied liberalization of control in natural resources in every country of Southern Africa taking the form of granting mining concessions, construction permits on tourist sites, land grant or concessions on communal land, timber permits and tourist operator agreements etc.

17.4 Conclusions

The results of the neo-liberal economic reforms show that the reform programmes have been a failure. But the nature of the failure can be understood better by taking a look at similar periods of time before and after the reforms and identifying basic trends if there are any. One such basic trend is that all the key economic indicators had been going down a long time before the reform program started. The reform programs did not create that trend. What the reform programs have done is to fail to reverse that trend.

Indeed one can say that in some areas there have been some improvements. Agricultural growth has thus been higher in the reform period than during the crisis period, but lower than the pre-crisis period, and in any case no higher than 2.7% far from adequate to reverse the trend and bring about transformation in agriculture. Thus reforms may have slowed the pace of deterioration of the African economic scene but they most definitely have failed to reverse the trend and bring about economic transformation. The same can be garnered from looking at GDP growth rates. While it is difficult to prove, it may be the case that all the figures would have been worse without the reforms.

Slowing down the rate of economic collapse is not a bad thing. But the fundamental issue is not to slow down the collapse but to bring about rapid growth and economic transformation. That has been the objective of the reform programs and that is how they should be evaluated. When evaluated from the point of view of rapid growth and transformation the reform programs have failed utterly.

African agriculture is demand constrained as a result of high transaction costs and low productivity and is stuck in a low growth trap. It has reached a dead end. Finance has utterly failed to perform its task of mobilizing savings and directing them towards profitable investments. It is caught in a trap of high real interest rates, high interest rate spreads, high incidence of non-performing loans, excess liquidity and low savings and low investments trap. Structural bottlenecks have thus led finance into a dead end. Industry has also reached a dead end because of the problems in agriculture and finance, because of infrastructural and institutional failures, etc. The economy as a whole is in a dead end.

Clearly African economies were in a dead end before the reforms for reasons that have to do with

the logic of the political economy in Africa and that of the predatory state based on it. The neoliberal reforms were intended to overcome that particular dead end. Instead they have led African economies to a new dead end. They have done so not because of technical problems but because of the fundamental nature of the reforms themselves.

The assumption behind the reforms was that while markets may be imperfect and African markets particularly so, African governments are even worse, that government failures is worse than market failure, and that markets would do better on their own however imperfect they might be, that in the end the markets will come through, work their magic and bring about economic transformation. The World Banks own analysis of the East Asian experience shows that there is no plausible theoretical basis for such an assumption. The African experience has shown that there cannot be an empirical basis for that assumption.

It is this assumption that precluded government investment in rural institutions, both of market supporting and technological capability accumulating type and trapped agriculture in a demand constrained circle. It is partly the logic of the reform which has crippled public investment in infrastructure in general and rural infrastructure in particular. It is the same logic which has precluded government intervention to address market failures in the capital and technology markets and other market failures and led finance and industry to a dead end. It is the fundamental logic of the reforms that has resulted in a new dead end.

The same logic also assumed that reducing the role of the government would be the primary instrument of reducing rent-seeking and hence overcoming the dysfunctional role of the predatory state in Africa. As we have shown in part one there is no reliable theoretical basis for making such an assumption. The practice in Africa has shown that rent-seeking continues unabated although in a different form. The reform program has also failed in transforming the rent-seeking political economy of Africa because its fundamental logic was flawed. It failed because you can not change a rent-seeking political economy by reducing the size and role of the state.

Thus while the reforms did not create Africa's economic crisis and may have slowed down its inexorable march to a catastrophe, it has led Africa to a new dead end rather that out of it. The economic failure is not simply about slow growth. It is about being in a poverty trap, being in a dead end. The economic failure is not about a specific policy instrument, it is about the fundamental logic of the neo-liberal paradigm and the Washington consensus variant of it. It is not a failure of economic perspective alone. It is a failure to transform a rent-seeking political economy too.

State intervention by a predatory state had miserably failed in Africa before the reforms. Returning to that failed paradigm is not going to take Africa out of the dead end. The Washington consensus has failed equally miserably in Africa in the post reform period. Maintaining that paradigm for another couple of decades is not going to take Africa of the new dead end. There has to be a third way, a third option. Theoretical analysis by some economists over the past two decades in particular has shown that there is indeed such a way. Experience in Taiwan and Korea has confirmed that such a third way can indeed work. A third way based on a developmentalist state appears to be the only way out.

18. Outcome of Political Reform

19. Some Country Stories

20. The African Renaissance and the Need for a Paradigm Shift

21.1 The Need for a Paradigm Shift

At the beginning of the 21st century there is renewed interest in Africa's economic and political destiny. Theories of the African renaissance have cropped up. Mechanisms for Africa to claim the 21st century have been suggested, including a program that has become the source of hope for many –NEPAD. All of these are indeed welcome developments. Many of these are, however, based on doing the things that Africans have been doing for the past two decades but doing them better. They are based, at best, on improving the neo-liberal paradigm that has governed Africa over the past two decades. The need for a Paradigm shift is not recognized in most of these programs. One is thus entitled to be skeptical about their outcome.

Africa has followed two paradigms since independence. The first one which held sway till the early 80s can be called the paradigm of the predatory state. The second one which has held sway since then is the neo-liberal paradigm. The African renaissance must be based on a thorough assessment of the two paradigms, and on analysis of whether the two paradigms have failed and a new paradigm shift is required.

The paradigm of the predatory state in Africa was based on: (a) a political-economy that is defined by pervasive rent-seeking; (b) a weak but large state, a "swollen state" based on such a political economy; and (c) an economy much of whose wealth outside the subsistence sector is controlled by the state and used for private wealth accumulation. This paradigm did bring about significant economic growth during the first decade of independence. But soon the growth potential of the paradigm was exhausted and Africa entered a dead end. While the external environment speeded up the process, it is the logic of the paradigm of the predatory state which led Africa to a dead end and a deep crisis. The predatory state was unable to reform itself and overcome the crisis. Indeed its very logic led it to feed on its decaying body and hence speed up its own dissolution rather than reform and transform itself. It had not only led Africa into the abyss but also proved that it was incapable of reforming itself and pulling Africa out of the abyss. The paradigm of the predatory state has been proved to be dysfunctional by the experience of the first two decades of independence.

The paradigm had thus to change. There was a clear need for a new paradigm. The shift in paradigm in Africa came as a result of the incapability of African states to reform themselves and the imposition of the then triumphalist neo-liberal paradigm on African governments. The shift in paradigm was also associated with an intellectual environment in the continent that was governed by what has been called the "There –Is- No- Other- Alternative" (TINA) syndrome. The neo-liberal

paradigm thus became practically and intellectually the only paradigm available, "the only game in town."

The results of the shift to the neo-liberal paradigm are clear and quite disappointing. The size of the state in Africa has been reduced while the state continues to be weak. The dissolution of the state has been slowed down and African states are increasingly plagued by what has been called the political economy of disorder. While the rates and scope of recovery have varied from country to country there has been a general economic recovery but there has not been sustained and rapid growth and transformation of the economy. African economies have entered a dead end path. While growth has stalled in many, a few are still growing and have not yet exhausted the limited growth potential within the essentially dead end path.

The reasons for such basic failure and limited results are and have been since the very beginning of the implementation of the neo-liberal paradigm hotly debated. In this regard it is instructive to note what an IMF economist had to say in the early 90s- "when the institutions are weak, the quality of governance poor, and the country's economy is very distorted, the results generated by programs are unlikely to be as desirable as one would like. But then the blame should not be placed on the programs. In such cases, nothing will work unless the type of governance is changed".

The economic crisis of the 80s was indeed the result of failure of governance, the result of a predatory state's logic. It was primarily a political problem. The solution to the economic crisis would, therefore, be primarily a political one, a change in governance. The neo-liberal paradigm rightly acknowledged that this was indeed the case and the reforms were designed to bring such a change in governance, to transform the predatory state, establish a benign night watchman state and thereby put Africa on a trajectory of sustained growth and transformation. While neo-liberal paradigm should not be blamed for creating the governance problem in the first instance it can and should be blamed for failing to bring about the desired change of governance, despite its clearly articulated program of bringing about such a change. On the fundamentals, the neo-liberal paradigm has been an unqualified failure in Africa.

The neo-liberal paradigm has failed to transform the state from a predatory to a benign one, from undemocratic to a democratic one in nearly two decades of trying to do so.

Transforming the predatory state in Africa would have required transforming the political economy on which it stood --- a political economy defined by pervasive rent-seeking. This would have meant changing the fundamental structure of incentives and disincentives of the economy from one that is inhospitable to productive activities to one that is inhospitable to rent-seeking activities.

In the case of the rural areas which are central to the politics and economics of Africa, it would have meant doing away with property relations that form the basis of vertical ties of patronage where they existed. It would have meant that the government would transform the incentives and disincentives in the rural economy in favor of productive activities and against rent-seeking activities by building the necessary infrastructure including roads and irrigation structures, address market failures in the rural credit, technology and goods markets including by building the necessary institutional infrastructure and promoting independent farmers organizations based on productive activities. It would have meant effective, extensive and well-designed intervention by a

strong state.

Transforming the political economy of rent-seeking would have also meant taking similar measures in the urban areas. These would have entailed restructuring the set of incentives and disincentives in favor of productive and against rent-seeking activities by extensive, effective intervention by a strong state to build the physical and institutional environment and to change the rules of the game.

The neo-liberal paradigm precludes the possibility of taking any of the measures mentioned above. The neo-liberal paradigm assumes that state action is the source of rent-seeking activity. It does not recognize the structural causes that create rent-seeking in developing countries. Extensive state intervention by any type of state would thus be considered a means of exacerbating the problem of rent-seeking rather than a means of overcoming it. An effective state intervention to change both the playing field and the rules of the economic game is thus not possible in an environment where the neo-liberal paradigm reigns.

Transforming the anti-democratic predatory state in Africa into a democratic one would have required the establishment of institutions of democratic pluralism based on production and productive activities. It would have meant promoting civic engagement and civic virtues through a thick network of horizontal associations. It would have meant eliminating or drastically reducing vertical networks of patronage. In view of the predominantly rural nature of Africa it would have meant creating a population of independent commercialized farmers, operating in a dynamically transforming agriculture and linked by a relatively dense network of independent farmers association and horizontal linkages. It would have in other words, required transforming the political economy of rent-seeking and patronage and building institutions of democratic pluralism based on productive activities and economic transformation.

By precluding the possibility of changing the playing field and rules of the economic game in favor of productive activities and economic transformation through effective state action, the neo-liberal paradigm precluded the establishment of dynamic agrarian democracy that could evolve overtime into mature urban based democracy. Instead the neo-liberal reform opted for what has been called trickle up democracy based on pluralism of patronage through the promotion of domestic NGOs and voluntary organizations as instruments of such patronage. Pluralism based on such patronage has continued to be plagued by the zero-sum game nature of politics based on rent-seeking and patronage. The democracy that is based on such a political economy has not only continued to be fragile but also still born in the sense that it has no tendency to evolve overtime into a mature urban based democracy.

The fundamental nature of the neo-liberal paradigm has thus led Africa into another economic dead end and into a fragile unstable democracy that is not only incapable of evolving into a stable and mature democracy but actually hinders the development of an alternative path of democracy that leads overtime to such a mature and stable democracy. The failure of the neo-liberal paradigm in Africa is thus not a failure of implementation of the paradigm or one of lack of refinement of this or that aspect of the paradigm. It is a paradigmatic failure and can only be corrected by a shift in paradigm.

Where the neo-liberal paradigm has failed to transform the political economy of rent-seeking the new alternative paradigm must be such as to successfully bring such a change and put Africa on a

trajectory of rapid and sustained growth and transformation. Where the neo-liberal paradigm has failed to transform the zero-sum game politics of Africa and establish a stable and sustainable democracy, the new alternative paradigm must lead to the establishment of healthy shoots of democratic pluralism of a broadly based agrarian democracy that is capable of evolving into a mature urban-based democracy. Once again the fundamental economic issue in Africa is a political one. It is an issue of transforming the pared down predatory state in Africa into a democratic and developmental one. As two African economists have rightly suggested the establishment of a democratic developmental state is the single most important task on the policy agenda in Africa.

What is required to bring about the African renaissance is thus clearly not the renaissance of the predatory state or modification of the neo-liberal paradigm. What is required is a paradigm shift towards the establishment of a democratic developmental state, a shift towards a paradigm which could be called democratic developmentalism.

21.2 The Possibilities for the Success of the Democratic Developmentalist Paradigm

Paradigm shifts in economic and political developments do not happen every day. They happen only when previous paradigms have failed and failed visibly, creating an environment of active search for alternatives. While the visible failure of an existing paradigm is necessary it is not sufficient. Developmental states like other states need a favorable internal and external environment to emerge and to succeed. And so if developmental states are to emerge in Africa, they will emerge and succeed not only because the neo-liberal paradigm has visibly failed but also because the global environment and the domestic environment are permissive of such a development, if not conducive to it.

The failure of the neo-liberal paradigm in Africa has meant the continued marginalization of Africa in an increasingly globalizing environment. Africa's share in global trade is negligible. Africa's share in global FDI is miniscule. When every part of the globe's wealth and well being is becoming increasingly dependent on everyone else's wealth and well being, a significant part of the globe is going out of circulation.

Companies in developed countries are looking for profitable investment opportunities across the globe. Taking Africa out of circulation means their choices are reduced, their profit making opportunities diminished. Bringing Africa into circulation, through the promotion of functional economies and dynamic market economies would clearly be in the interest of such companies. Companies in developed economies are looking for new and larger markets to sell their products and services. With Africa out of circulation due to its dysfunctional economy and abject poverty the availability of larger and profitable markets to sell products and services in gets reduced. Dynamic African economies with significant buying powers would thus be in the clear interests of such companies.

In the end, therefore, the basic economic interests of the developed countries do not lie in the perpetuation of dysfunctional economies and polities in Africa. Their basic economic interests lie in dynamic, rapidly growing market economies in Africa that can then provide profitable opportunities for large amounts of FDI and sizeable markets for their companies. As democratic developmentalist states would create dynamic, rapidly transforming market economies in Africa,

such development would not only pull the peoples of Africa out of misery and remove the specter of failed states from Africa, but would also be consistent with the fundamental interests of the developed countries. As the permissive attitude of the major powers if not their enthusiastic support is vital for the emergence and success of democratic developmental states in Africa, the fact that such development is consistent with their basic economic interests means that it is likely that the external environment will at least be permissive for the emergence and success of democratic developmentalist states in Africa.

The increasing globalization of the world, a world still based on sovereign states means that what happens in Africa affects everyone including the major powers. African states are no longer important as diplomatic clients, as they were during the Cold War, or at least the significance of such diplomatic clientage has decreased dramatically. But in the era of globalization and the political economy of disorder that is becoming prevalent in Africa, African countries are important as sources of transnational security threats. It is in the security interests of the major powers to prevent the dissolution of African states.

Such interests are becoming clearer by the day. The political economy of disorder which creates a fertile breeding ground for organized crime of all types from drug trafficking to money laundering seriously affects the most developed of countries. Unless the swamp that breeds such risks is dried at source the threat posed by organized crime based in Africa to the developed world will increase. Killer diseases are emerging from time to time and are spreading throughout the globe quickly. A black hole of dysfunctional polities in the geographic middle of the globe would enormously increase the risk posed by such diseases and undermine the attempts to control and manage them.

Failed and failing states in Africa are generating refugees by their hundreds of thousands knocking at the doors of the developed countries. Such emigration could easily multiply by a factor of ten if the major countries in Africa continue to inch towards state failure. Immigrants are already posing a serious economic, social and political challenge to these countries. A scenario in which the phenomenon would increase say ten fold would indeed be a nightmare scenario from the point of view of the developed countries. Putting a lid on such a development would therefore be clearly in their national security interest.

As the events of September 11 have shown terrorism has become one of the key emerging threats of the 21st century. A combination of massive improvement in technology and communication, populations that are excluded and hence aggrieved by their condition, failed or failing states, fanatics committed to wreck havoc and increasing possibilities of massively damaging linkages between terrorism and weapons of mass destruction and information warfare has become a lethal threat to everyone including the developed countries. No part of the globe has the level of dysfunctionality in politics and economics that Africa has --Africa is thus likely to increasingly become the base of such terrorism.

Democratic developmental states would be able to tackle all of these problems effectively because they would create dynamic and functional economies and polities. In so doing they will not only serve the interests of African peoples, but would also give maximum contribution to address the security concerns of everyone else. It would thus be in the security interest of developed countries to create a permissive if not a conducive, environment for developmental states to emerge and to

succeed in Africa.

The global environment is also permissive now because the neo-liberal paradigm is no longer the unchallenged and triumphalist paradigm that is was a decade or even five years ago. It has visibly failed in Africa. Alternatives to it have been suggested from different quarters. The intellectual environment is thus permissive to alternatives to the neo-liberal paradigm such as the democratic developmentalist paradigm.

Very few developmental states have been democratic. In Africa, the external circumstances are such that a developmentalist state will also have to be a democratic one. African countries are being forced to democratize not only because of the appeal of democracy as a form of governance but also because it has become a condition for getting adequate policy rent. Perhaps more importantly, in an age of the ascendancy of democracy ethnically diverse African countries cannot continue to exist unless they have some democratic legitimacy. The Cold War environment which had made undemocratic unity possible is no longer there and as one writer has put it "There is no way in which an artificial African state can hold together unless most of its people want to. The choice, then is between governments that can achieve some popular legitimacy, whether or not these are multi-party democracies in a Western liberal sense of the term, and the collapse of ordered states into anarchy".

A developmentalist state would by definition have to transform the political economy of patronage and rent-seeking and hence the very basis of the zero-sum politics in Africa. That, in itself, would be fundamental to establishing viable democracies in Africa. It would also have to build national consensus around its development project, make its "development project hegemonic." Democracy facilitates that. Thus the circumstances for a developmental state to also be a democratic one are much more conducive in present-day Africa than they have been in the past elsewhere.

The external environment is, however, not rosy from the point of view of democratic developmentalism. The rent-seeking coalition that has kept Africa down is not limited to African rent-seekers. There are bound to be some in the "development business" who have a stake in the current state of affairs. There are developed country companies who live off the opportunities created for them by the development aid relationship. There are developed country companies who thrive on rents from mining the natural resources of Africa. Perhaps more importantly there are sectors of society in such countries who would lose if preferential and non-reciprocal access to their markets for Africa were to become a practical reality. Such access has played a key role in the success of developmental states in Taiwan and Korea and cannot but play the same role in Africa. Overcoming such vested rent-seeking interests is likely to become a very significant hurdle for the emergence and success of developmental states in Africa.

Perhaps a more difficult challenge would be that which is posed by intellectual conservatism. Too many people in the advanced countries, too many institutions have staked their reputation on the neo-liberal paradigm to be able to change easily to a new paradigm. Indeed even those who have no such stake are unlikely to change their views in favor of what is likely to appear a mere hypothetical possibility. Others are likely to interpret the challenge to the neo-liberal reforms in Africa as a challenge to neo-liberalism in general. The fact that neo-liberalism in the advanced countries is based on a different political economy, one that is conducive to value and wealth creation, and thus does not have the same malignant impact as it has had in Africa is unlikely to be

enough of a consolation to them. In this sense the challenge of intellectual conservatism may be an ever higher and tougher hurdle to overcome than that of vested interests.

Thus while the external environment is at least permissive and possibly conducive to the emergence and success of a democratic developmentalist paradigm in Africa, it is also full of tough and high hurdles that need to be overcome if there is going to be a paradigm shift and a real as opposed to rhetorical African renaissance.

The domestic scene in African countries is also similarly permissive. It is clear that many African countries are failing or potentially failing states. Their economies and politics has entered a dead end. Their political economy of rent-seeking is fast manifesting itself as the political economy of disorder. The paradigms of development that most sub-Saharan African countries have followed have failed and failed disastrously. Such circumstances are precisely the circumstances, which encourage people to look for alternative solutions alternative paradigms as a means of national renaissance and survival. To the extent that the circumstances require a different paradigm to prevent dissolution of African states and polities and the chaos that has gripped the failed and failing African states, the domestic environment in Africa can be considered to be conducive to the emergence and success of the paradigm of democratic developmentalism.

Many African countries have small farmer dominated agricultural sectors and communal land ownership systems. Patronage and rent-seeking in much of rural Africa has thus to do more with rural marketing and state power than landownership system. Removing vertical linkages of patronage from the rural areas as a key initial step in the democratic developmentalist paradigm does not require difficult land reform in much of Africa. Political reforms to remove the predatory state and overcome rural marketing problems would suffice in most African countries. The circumstances in rural Africa are thus conducive to the emergence and success of a democratic developmentalist version of "the rural strategy".

At the same time there is no question that there are very daunting challenges that face the new paradigm. African peoples have lived in dysfunctional political economies of rent-seeking and patronage. The only politics they know is the zero-sum politics of patronage and the use of instruments such as ethnicity for purposes of patronage. The only concept of "development" they know is development that is done to them, the development of patronage and "pulling projects". That is a paradigm which defines development as a process that is done by the people rather than done to them, a process of pulling the resources and creative energies of the people rather than "pulling projects" through patronage from NGOs and donors is likely to at least initially face a skeptical populace. A paradigm designed to promote civic engagement through horizontal networks and associations, to promote civic virtues, and democratic pluralism including ethnic and religious pluralism is likely to face a skeptical audience.

There are vested interests who benefit from the current dysfunctional system that are likely to vigorously oppose the new paradigm. The rent-seeking businesses, the "ethnic entrepreneurs" who have used ethnicity as a means of accessing state power and accumulating personal wealth, many of the domestic "NGO and VO entrepreneurs" who have used domestic NGOs and voluntary organizations as instruments of patronage and personal welfare, if not wealth accumulation, are likely to at least initially vigorously oppose the democratic developmentalist paradigm.

On the other hand, the fact that most African countries are democracies of sorts, however fragile their democracy might be, creates the political space for political activists to embrace the new paradigm and mobilize political support for it. Both the domestic and external environment are thus permissive if not conducive to the emergence and success of the democratic developmentalist paradigm.

An overall assessment of the conditions in Africa suggests that it is possible for democratic developmentalists states to be established in Africa. In this regard what is crucial from a continental point of view is that a significant number of countries establish such a state and succeed. This could then have a flying geese effect on the rest of the continent. One can for example imagine the impact that a successful developmental state in Nigeria might have throughout West Africa and across the continent.

21.3 Steps in the Direction of the African Renaissance.

If the 21st century is to be an African century, a century of African renaissance there has to be a paradigm shift in Africa's economic and political development in the direction of democratic developmentalism. Developmental states come in all shapes and sizes and therefore there cannot be a single blue print for all democratic developmentalist states in Africa. Indeed not all African states can become developmental ones. Hence democratic developmentalism cannot be the single blue print for every country in the continent. It is, however, possible to identify a few general points that can lead to the emergence and success of developmentalist states in Africa, enough such successes to have profound impact on the destiny of the continent as a whole. These are as follows.

- a) Overcome TINA (there is no alternative). In the end leadership is bound to play a critical role in exploiting the permissive environment to make democratic developmental states happen in Africa. Such leadership must be able to articulate an alternative vision and convince enough people to be able to win elections and transform the predatory African state. The "there is no other alternative" syndrome has to be broken if such a democratic developmental movement is to be created. African intellectuals must break the taboo and challenge all received wisdom; debates on all alternative avenues for national and continental renaissance must be kindled. Such a debate should not be aimed at replacing the TINA syndrome of the neo-liberal paradigm with a syndrome of the democratic developmentalist variety but at providing various alternative paths.
- (b) *Create a Global Developmentalist Coalition*. Democratic developmental states in Africa are going to need all the help they can get to emerge and succeed. Such help is needed in the form of development aid. In this regard the key thing is unlikely to be the amount of assistance, because the amount of such aid that Africa gets is bigger than what Taiwan and Korea got to get them started. It is true that in terms of infrastructure the Japanese had done a lot more in these countries than Africa has managed so far and thus Africa including developmental states in Africa may need the higher level of assistance they are getting. The key thing is, however, going to be a drastic change in its quality.

A change in its quality implies that at least in developmental states the focus of such assistance must be on building infrastructures particularly rural infrastructure including rural roads and

irrigation and water harvesting facilities. It should also focus on building the capacity of the state. A change in the quality would also mean that aid should be dispensed on the basis of a clearly set developmental program and as a supplement to the efforts of the people rather than as patronage to be distributed piecemeal through networks of patronage, both state and non-state.

African countries face huge shortfalls in physical and institutional infrastructures, encounter massive market failures and are therefore unable to compete globally on purely free trade terms. They need to export while at the same time using trade policy to support and encourage infant industries. They need to attract productive FDI to accelerate their development and need to provide various incentives to compensate for the various shortfalls mentioned above. In other words they need non-reciprocal access to developed country markets. Guaranteed long-term access to developed country markets on what has sometimes been called, special and differential treatment terms is thus perhaps the key assistance that African states need. That is the type of market access that Taiwan and Korea were given by the United States.

Help is needed most particularly in creating a conducive global trading environment for developmental states in Africa to succeed. African states do not have the institutional and physical infrastructure to compete on a global free-market. A privileged and non-reciprocal access to developed markets is thus critical for their success. They need such access to compensate for their institutional and infrastructural weakness and attract domestic and foreign investment of the productive type, which in turn would generate the resources and experience required to overcome the institutional and infrastructural failures. They need non-reciprocal access to such markets also to be able to use various instruments of industrial policy to overcome the pervasive market failures. They cannot do so if they are asked to reciprocate for the access they get to developed markets. Thus an appropriate and predictable special and differential treatment of Africa for a reasonably long period is bound to be perhaps the most important aid that developmental states in Africa are going to need.

While developed countries have been willing to give Africa unprecedented levels of policy rent, there has been reluctance on the one key form of aid that will be crucial for the success of developmental states in Africa –access to their markets. This has led some writers to state "If industrialized countries are sincere in their support for diversification of African economies and in their demands for a secure environment for foreign investment, they should provide a guarantee of market access for the non-traditional exports of African countries. The ideal would be a commitment that no African exports would be subject to non-tariff or tariff barriers for an extended period say fifteen years." It is such aid that developmental states in Africa are going to require.

There has to be more political space for experimentation in development policy than has been the case so far in Africa. A space to make the choiceless democracy that has been the result of trickle up democracy in Africa6, one which offers real choices and alternatives must be created. The international community has a role in creating such a space by tolerating development paradigms that are different from the orthodoxy preached by it. Africans have to demand and create such a space.

The global developmentalist coalition that Africa must establish cannot and must not be limited to governments and multilateral institutions. Indeed the most steadfast external constituency for the

democratic developmentalist paradigm in Africa is likely to be non-governmental. Progressive political groups, non-governmental organizations, activists and progressive intellectuals of various types are already engaged in various activities that are supportive of Africa's renaissance. Consolidating and re-energizing such as external constituency is likely to be a key step in establishing a global developmentalist coalition for Africa's renaissance.

(c) *Create a solid domestic developmental coalition.* The success or failure of the democratic developmentalist paradigm will depend on the strength and breadth of developmental coalition created in each country. The critical factors keys to the creation of such coalition are breaking the TINA syndrome and establishing a solid developmental base in the rural areas.

The rural areas are crucial to the success of democratic developmentalism for a number of reasons. Rapid and sustained development in African countries will crucially depend on agriculture. It is the dynamism of agriculture which is going to determine the dynamism of the non-agricultural sector and the economy as a whole and hence determine the success of the project as a whole. Democracy in Africa has to be an agrarian one if it is to be stable and capable of evolving into a mature urban based one. Hence the political and economic aspects of the project crucially depend on solid grass-roots base in the rural areas.

Establishing such a solid democratic developmentalist base in the rural areas is going to require dismantling patronage networks, promoting the establishment of a dense network of independent farmers organizations and associations based on production and making sure that external assistance supplements the developmental effort of such associations rather than overwhelming them in an environment of patronage. It will also involve the single minded focus on transforming rural Africa from subsistence farming to commercialized and technologically dynamic farming, and making the necessary investment in infrastructural and institutional development.

The developmental state in Africa will have to base its developmental coalition firmly on the rural areas but not limit it to it. It will have to keep its independence from the private sector while bringing them around to play the developmental game on the basis of a new playing field and new rules. It will have to convince the urban population at large that democratic developmentalism is indeed the path of national renaissance. In the end its actual successes will be as important if not more so as its articulation of a new vision in cementing the developmental coalition.

(d) Rebuild the African State. While the specific policies that need to be designed to address market failures are necessarily going to vary from country to country the need for doing so through effective government intervention will always be there. Government intervention to restructure the set of incentives and disincentives in favor of productive investment and against rent-seeking by, among other things, addressing the pervasive market failures that inhibit productive investment is bound to be a central and distinguishing characteristic of the developmental state.

Such effective government intervention cannot happen with the "swollen, and weak essentially antidemocratic state. A fundamental transformation and building up of the state is thus bound to be a central tenet of the paradigm. The state will have to be strengthened primarily by transforming it from the locus of personal wealth accumulation to an effective instrument of restructuring the playing field and enforcing the new rules of productive investment. That will require a change in outlook and attitude among state personnel, the creation of a new developmental ethics within the state, and the acquisition of the necessary skills to deliver the required service. It will also require major restructuring to increase its efficiency and effectiveness.

State transformation will also require building solid state structures in the rural areas based on the grass-roots activism of the small farmers and their independent associations. The main locus of action at least initially is going to be in the rural areas and hence a strong presence of the state based on its developmentalist coalition and support is going to be key to rebuilding the state.

State rebuilding is also going to require establishing new institutions, restructuring old ones so that the state will have the necessary means to intervene effectively in addressing market failures.

Rebuilding the state as part of the implementation of the democratic developmentalist paradigm must also include launching the state on a trajectory of vibrant and dynamic democracy. Transforming the rent-seeking political economy will eliminate the zero-sum politics associated with it and hence create the basis for sustainable democratization. Dismantling networks of patronage particularly, in the urban areas, promotion of independent farmers organizations based on production rather than patronage is a key to establish a stable agrarian democracy in Africa. Rebuilding the legislature, judiciary and executive along democratic lines, inculcating democratic values through civic education and civic engagement via associations of horizontal linkage are also essential elements of transforming and rebuilding the African state.

BIBLIOGRAPHY (to be completed)

Abdi Ismail Samatar. An African Miracle: State And Class Leadership And Colonial Legacy in Botswana Development.

Adebayo O. Olukoshi. "The Elusive Prince of Denmark: Structural Adjustment And The Crisis of Governance in Africa", African Voices on Structural Adjustment.

Aksel De Lasson. The Farmers Association Approach To Rural Development- The Case of Taiwan. Alan Fowler. "The Role of NGO's in Changing State. Society Relations: Perspectives From Eastern And Southern Africa", Development Policy Review volume 9, 1991.

Albert Park And Bruce Johnston, "Rural Development And Dynamic Externalities in Taiwan's Structural Transformation" Economic Development And Cultural Change volume 44, 1995.

Alexander Gerschenkron. Economic Backwardness in Historical Perspective; A Book of Essays.

Alice H. Amsden. Asia's Next Giant: South Korea And Late Industrialization.

Alice H.Amsden. The Rise of The "Rest" Challenges to the West From Late Industrializing Economies.

Alice H. Amsden. "The State And Taiwan's Economic Development", Bringing The State Back in.

Alice H. Amsden. "Taiwan's Economic History: A Case of Etatisme And a Challenge to Dependency Theory" Modern China volume 5 No 3, 1979.

Alvin Y. So And Stephen W.K. Chiu. East Asia And The World Economy.

Anna Simmons. "Somalia: The Structure of Dissolution" The African State at a Critical Juncture:

Between Disintegration and Reconfiguration.

Anthony J. Regan. "Decentralization Policy: Reshaping State And Society" Developing Uganda.

Anthony Y.C. Koo. The Role of Land Reform in Economic Development: A Case Study of Taiwan.

Arne Bigsten And Steve Kayizzi. Mugerwa. "Is Uganda an Emerging Economy? Nordiska Africanstitutet, Research Report No 118.

Aswantic Saith. "Contrasting Experiences in Rural Industrialization: Are the East Asian Success Transferable" Rural Industrialization And Employment in Asia.

Awishay Braverman And J. Luis Guash. "Administrative Failures in Rural Credit Markets", The Economics of Rural Organization, Theory, Practice And Policy.

Barrington Moore. Social Origins of Dictatorship And Democracy: Lord And Peasant in The Making of The Modern World.

Benedict Stavis. Rural Local Organizations And Agricultural Development in Taiwan.

Bent Dalum, Bjorn Johnson And Bingt Ake Lundervall. "Public Policy in The Learning Society"

National Systems of Innovation: Towards a Theory of Innovation and Interactive Learning.

Bernard Weimer And Olaf Claus. "A Changing Southern Africa: What Role for Botswana?" Botswana: The Political Economy of Democratic Development.

Bjorn Johnson. "Institutional Learning" National Systems of Innovation: Towards a Theory of Innovation And Interactive Learning.

Cal Clarck. Taiwan's Development: Implications For Contending Political Economy Paradigms. Catherine Boone. "The Making of a Rentier Class: Wealth Accumulation And Political Control in Senegal" Africa Dilemmas of Development And Change.

Chalmers Johnson. "Political Institutions And Economic Performance The Government-Business Relationship in Japan, South Korea, And Taiwan", The Political Economy of the New Asian Industrialism.

Charles Harvey. "Banking Policy in Botswana" Banking in Africa: The Impact of Financial Sector Reform Since Independence.

Charles O. Chikeka. Britain France And The New African States: A Study of Post Independence Relationships 1960-1985.

Christopher Clapham. Africa And The International System.

Christopher Colcough And Stephen McCarthy. The Political Economy of Botswana: A Study of Growth And Distribution.

Christopher Delgado. "The Context And Rational for Market Reforms" Reforming Agricultural Markets in Africa.

Christopher Delgado. "Agricultural Transformation: The Key to Broad-based Growth And Poverty Alleviation in Africa" Agenda for Africa's Economic Renewal.

Cynthia Taft Moris And Irma Adelman. "Nineteenth Century Development Experience And Lessons For Today" The Selected Essays of Irma Adelman volume I.

Daniel Green. "Ghana Structural Adjustment And State (Re)formation" The African State at a Critical Juncture: Between Disintegration And Reconfiguration.

Dani Rodrick. "Getting Interventions Right: How South Korea And Taiwan Grew Rich." NBER Working Paper # 4964, 1994.

Dani Rodrick. "Political Economy And Development Policy" European Economic Review volume 36, 1992.

Dani Rodrick. "Trade And Industrial Policy Reform" Hand Book of Development Economics volume 3B.

David C. Collander. "Introduction", Neo-Classical Political Economy: The Analysis of Rent-seeking And DUP Activities.

David Kaimowitz. "Preface", Economic Change, Governance & Natural Resource Wealth: Political Economy of Change in Southern Africa.

David K. Leonard And Scott Strauss. Africa's Stalled Development: International Causes And Cures.

David Reed. "The Political Economy of Natural Resource Wealth" Economic Change, Governance & Natural Resource Wealth: Political Economy of Change in Southern Africa.

David Reed And Ola Larson. "Natural Resource Wealth in The Construction of Neo-Liberal Economies in Southern Africa" Economic Change, Governance & Natural resource Wealth: Political Economy of Change in Southern Africa.

Deborah Brautigam. "State Capacity And Effective Governance. The State as Agent: Industrial Development in Taiwan 1952-1972" Asian Industrialization And Africa: Studies in Policy Alternatives to Structural Adjustment.

Denis Fred Simon. "Taiwan's Strategy for Creating Competitive Advantage: The Role of the State in Managing Foreign Technology" Taiwan's Enterprises in Global Perspective.

Dieter Ernest, Tom Ganiatsos And Lynn Mytelka. "Technological Capabilities in the Context of Export-led Growth- a Conceptual Framework", Technological Capabilities And Export Success in Asia.

Dietrich Rues Chemeyer And Peter Evans. "The State And Economic Transformation: Toward an Analysis of The Conditions of Effective Intervention" Bringing The State Back in.

Dimitri Vittas And Yoon Je Cho. "Credit Policies, Lessons From Korea And Japan" World Bank Research Observer volume II No 2.

Dwight H. Perkins And Michael Roemer. "Differing Endowments And Historical Legacies "Asia And Africa: Legacies And Opportunities in Development.

Douglas C.North. Institutions, Institutional Change And Economic Performance.

Eboe Hutchful. Ghana's Adjustment Experience: The Paradox of Reform.

Eleni Z. Gabre-Medhin And Bruce Johnson. "Accelerating Africa's Structural Transformation: Lessons from East Asia" MSSD Discussion Paper No 34, 1999.

E. Gyimah Boadi And Nicolas Van De Walle. "The Politics of Economic Renewal in Africa" Agenda For Africa's Economic Renewal.

E.Gyimah Boadi And Richard Jeffries. "The Political Economy of Reform" Economic Reforms in Ghana: The Miracle And the Mirage.

Eul Yong Park. "Foreign Trade, Direct Investment And Industrialization Strategy of Korea" KDI Discussion Papers 1991.

Jeoffrey B. Tukahebura. "Privatization As a Development Policy" Developing Uganda.

Giovanni Dossi And Luc Soete. "Technical Change And International Trade" Technical Change And Economic Theory.

Gloria Somolekae. "Bureaucracy And Democracy in Botswana: What Type of Relationship", Botswana, The Political Economy of Democratic Development.

Gustav Ranis. "Another Look at the East Asian Miracle" World Bank, Economic Review volume 9 No 3.

Harry T. Oshima. "The Transition From An Agricultural to an Industrial Economy in Asia" Economic Development And Cultural Change volume 34, 1986.

Henry Ergas. "The Importance of Technology Policy," Economic Policy And Technological Performance.

Hernando De Sotto. The Mystery of Capital: Why Capitalism Triumphs in The West And Fails Everywhere Else.

Howard Pack. "The Role of Exports in Asian Development: Technological Change And Growth in

East Asia Macro versus Micro Perspectives" Rethinking the East Asian Miracle.

Hsin-Huang Michael Hsiao. Government Agricultural Strategies in Taiwan And South Korea: A Macro Sociological Assessment.

Ian Livingstone. "Developing Industry in Uganda in 1990s" Developing Uganda.

Irma Adelman And Steven J. Vogel. "The Relevance of ADLI for Sub-Saharan Africa", The Selected Essays of Irma Adelman volume I.

Jack Parson. "The Labor Reserve" in Historical Perspective: Toward A Political Economy of Bechuanaland Protectorate." The Evolution of Modern Botsawana.

Jack Parson. "Liberal Democracy, The Liberal State And The 1989 General Elections in Botswana" Botswana, The Political Economy of Democratic Development.

Jack R. Andersen And Karla Hoff. "Technological Change, Imperfect Markets And Agricultural Extension: An Overview" The Economics of Rural Organization, Theory, Practice And Policy. James N. Buchanan. "Rent-seeking And Profit Seeking" Toward A Theory of The Rent-seeking Society.

Jang Sup-Shen. The Economics of the Late Comers: Catching Up, Technology Transfer And Institutions in Germany, Japan And S. Korea.

Jean Francois Bayart. The State in Africa: The Politics of the Belly.

Jean Francois Bayart, Stephen Ellis And Beatrice Hibou. The Criminalization of The State in Africa. Jeffrey Herbst. "The Politics of Sustained Agricultural reform in Africa" Hemmed in: Responses To Africa's Economic Decline.

John D. Holm. "Botswana A Paternalist Democracy" Democracy In Developing Countries volume II Africa.

John Landsdale. "Political Accountability in African History" Political Domination in Africa: Reflections on the Limit of Power.

John Ravenhill. "A second Decade of Adjustment, Greater Complexity Greater Uncertainty" Hemmed in: Responses to Africa's Economic Decline.

John Toye. "Structural Adjustment: Context, Assumptions Origin and Diversity" Structural Adjustment And Beyond in Sub-Saharan Africa.

John W. Mellor. "Introduction" Agriculture on the Road to Industrialization.

Jose Edgardo Campos And Hilton L. Root. The Key to the Asian Miracle: Making Shared Growth Credible.

Joseph A. Yager. Transforming Agriculture in Taiwan: The Experience of The Joint Commission on Rural Reconstruction.

Joseph E. Stiglitz. Wither Socialism.

Joseph E. Stiglitz "Some Lessons From The East Asian Miracle" World Bank, Research Observer volume II No 2.

Joseph E. Stiglitz. "Participation And Development, Perspectives From The Comprehensive Development Paradigm" Joseph Stiglitz And The World Bank: The Rebel Within.

Joseph E. Stiglitz. "Markets, Market Failures And Development" Persepectives on Economic Development volume 79 No 2.

Joseph E. Stiglitz And Marilou Uy. "Financial Markets, Public Policy And The East Asian Miracle" The World Bank, Research Observer volume II No 2.

Joseph E. Stiglitz And Carl E. Walsh. Principles of Micro Economics.

Joseph E. Stiglitz And Karla Hoff. "Imperfect Information And Rural Credits, Puzzles And Policy Perspectives" The Economics of Rural Organization, Theory, Practice And Policy.

J. Stephen Morrisson. "Botswana's Formative Late Colonial Experiences" Botswana: The Political Economy of Democratic Development.

J.L. Enos And W.H. Park. The Adoption And Diffusion of Imported Technology: The Case of Korea. Jung En. Woo. Race to the Swift: State And finance in Korean Industrialization.

Justin Yifu And Yang Yoo. "Chinese Rural Industrialization in The Context of the East Asian Miracle." Rethinking the East Asian Miracle.

Kazuo Koike. "The theory of Skill Formation System" Skill Formation in Japan And South East Asia.

Klaus Schmidt Hebbel, Luis Serven And Andrias Solimano. "Saving And Investment: Paradigms, Puzzles, Policies" The World Bank, Research Observer volume II No 1.

Kwan S. Kim. "The Korean Miracle (1962-80) Revisited: Myths And Realities in Strategies And Development." Asian Industrialization And Africa: Studies in Policy Alternatives to Structural Adjustment.

Larry Diamond. "Roots of Failure, Seeds of Hope." Democracy in Developing Countries volume II Africa.

Larry E. Westphal, Linsu Kim And Carl S. Dahlman: "Reflections On The Republic of Korea's Acquisition of Technology", International Technology Transfer.

Leroy P. Jones And II Su Kong. Government, Business And Entrepreneurship in Economic Development- The Korea Case.

Lewis A. Kasekende. "Capital Account Liberalization: The Ugandan Experience" Development Policy Review volume 19 No 1 2001.

Lewis A. Kasekende And Michael Atingi Ego. "Impact of Liberalization on Key Markets in Sub-Saharan Africa, The Case of Uganda" Journal of International Development 11, 1999. Linsu Kim. Imitation to Innovation.

Louis T. Picard. "From Bechuanaland to Botswana: An Overview" The Evolution of Modern Botswana

Lynn Mytelka. "Stimulating Effective Technology Transfer: The Case of Textiles in Africa" International Technology Transfer.

Lynn Mytelka And Dieter Ernest. "Catching up, Keeping Ahead. The Korean Model Under Pressure" Technological Capabilities And Export Success in Asia.

Machiko Nissanke. Financing Enterprise Development And Export Diversification in Africa" UNCTAD, African Development in A Comparative Perspective, Study No 8.

Machiko Nissanke And Ernest Aryelty. Financial Integration And Development: Liberalization And Reform in Sub-Saharan Africa.

Martin Brown Bridge. "Financial Repression And Financial Reform in Uganda," Banking in Africa: The Impact of Financial Sector Reform Since Independence.

Martin Brown Bridge And Augisine Fritz Gockel. "The Impact of Financial Sector Reform in Ghana," banking in Africa: The Impact of Financial Sector Reform Since Independence.

Martin Brown Bridge And Charles Harvey. "Conclusions" Banking in Africa: The Impact of Financial Sector Reform Since Independence.

Massoud Karshenas. "Capital Accumulation And Agricultural Surplus in Sub-Saharan Africa And Asia" UNCTAD, African Development in A Comparative Perspective, Study No 1.

Michael E. Porter. The Competitive Advantage of Nations.

Michael P.Todaro. Economic Development 6th Edition.

Michael Savel. "Growth in East Asia What We Can And What We Cannot Infer" IMF, Economic Issues 1.

Michael S. McPherson. "Limits on Self-seeking: The Role of Morality in Economic Life", Neo-Classical Political Economy: The Analysis of Rent-seeking And DUP Activities.

Michael Twaddle and Holger Bernt Hansen. "The Changing State of Uganda", Developing Uganda.

M. Shahid Alam. Governments And Markets in Economic Development Strategies: Lessons From Korea, Taiwan And Japan.

Mylene Kherallah. "Introduction", Reforming Agricultural Markets in Africa.

Mylene Kherallah. "Conclusions And Policy Implications," Reforming Agricultural Markets in Africa.

Nancy Birdsall, David Ross & Richard Sabot. "Education, Growth And Inequality". Pathways To Growth: Comparing East Asia And Latin America.

Nancy Birdsall, David Ross & Richard Sabot. "Inequality And Growth Reconsidered: Lessons From East Asia," World Bank, Economic Review volume 9 No 3.

Naomi Chazan And Donald Rothchild. "The Political Repercussions of Economic Malaise" Hemmed in: Responses to Africa's Economic Decline.

Nicolas Van De Walle. African Economies And The Politics of Permanent Crisis 1979-1999. N. Gregory Mankiw. Macro Economics.

Norman T. Uphoff And Milton J. Esman. Local Organizations For Rural Development: Analysis of Asian Experience.

OECD. Reviews of National Policies for Education: Korea.

Padraic White. "How The IDA Operates" The Making of the Celtic Tiger: The Inside Story of Ireland's Boom Economy.

Park Chung Hee. The Country, The Revolution And I.

Park Jin Hwan. The Saemaul Movement: Korea's Approach to Rural Modernization in the 1970s. Patrick Chabal. Power in Africa.

Patrick Chabal. "The Quest For Good Government And Development in Africa is NEPAD The Answer" International Affairs volume 78, No 3 July 2002.

Patrick Chabal And Jean Pascal Daloz. Disorder As Political Instrument.

Patrick M. Molutsi. "International Influences on Botswana is Democracy" Botswana: The Political Economy of Democratic Development.

Paul M. Romer. "Endogenous Technological Change," Journal of Political Economy volume 98 No 5.

Paul Winters, Alan de Janvry, Elizabeth Sadoulet And Kostav Stamoulis. "The Role of Agriculture in Economic Development: Visible And Invisible Surplus Transfers" World Bank, Working Paper No 814.

Peter Evans: "The State as Problem And Solution: Predation, Embedded Autonomy And Structural Change." The Politics of Economic Adjustment.

Peter Lewis. "Political Transition And The Dilemma of Civil Society in Africa". Africa Dilemmas of Development And Change.

Peter Lewis And Howard Stein. "Shifting Fortunes: The Political Economy of Financial Liberalization in Nigeria." World Development volume 25 No 1, 1997.

Phillip A. Huxtable. "The African State Toward The Twenty First Century: Legacies of The Critical Juncture" The African State at a Critical Juncture: Between Disintegration And Reconfiguration.

Philippe Aghion And Peter Howitt. Endogenous Growth Theory.

Ray Mac Sharry And Padraic White. "From Celtic Pauper to Celtic Tiger" The Making of the Celtic Tiger, The Inside Story of Ireland's Boom Economy.

Reginald Herbold Green. "The IMF And The World Bank in Africa: How Much Learning" Hemmed in: Responses to Africa's Economic Decline.

Richard Sandbrook. The Politics of Africa's Stagnation.

Robert A. Dahl. Democracy And Its Critics.

Robert Fatton Jr. Predatory Rule: State And Civil Society in Africa.

Robert K. Hitchcock. "Water, Land And Livestock: The Evolution of Tenure And Administration

Patterns in the Grazing Areas of Botswana" The Evolution of Modern Botswana.

Robert Putnam, Robert Leonardi and Rafaela Y. Nanetti. Making Democracy Work: Civil Traditions in Modern Italy.

Robert Wade. Governing The Market: Economic Theory And The Role of Government in East Asian Industrialization.

Robert Wade. "The Role of Government in Overcoming Market Failures: Taiwan, Republic of Korea And Japan" Achieving Industrialization in East Asia.

Roger Tangri And Andrew Mwenda. "Corruption And Cronyism in Uganda's Privatization in 1990s" African Affairs, 100. 2001.

Roger Yeager. "Governance And Environment in Botswana: The Ecological Price of Stability" Botswana: The Two Political Economy of Democratic Development.

Samuel P. Huntington. The Third Wave: Democratization in The Late Twentieth Century.

San Gee And Wen Jeng Kuo. "Export Success And Technological Capability: Textiles And

Electronics in Taiwan Province of China" Technological Capabilities And Export Success in Asia.

Stephen Haggard. "The Politics of Industrialization in The Republic of Korea And Taiwan" Achieving Industrialization in East Asia.

Stephen Haggard. Pathways From The Periphery: The Politics of Growth in Newly Industrialized Countries.

Stephen Haggard And Robert R. Kaufman. The Political Economy of Democratic Transitions.

Stephen Haggard And Robert R. Kaufman. "Economic Adjustment And The Prospects for Democracy" The Politics of Economic Adjustment.

Stephen Haggard And Tun-Jen Cheng. "State And Foreign Capital in the East Asian NICS," The Political Economy of the New Asian Industrialism.

Sung Kwan Ban, Pal Yong Moon And Dwight H. Perkins. Studies in the Modernization of the Republic of Korea 1945-1975. Rural Development.

Susan Dicklitch. The Elusive Promise of NGOs in Africa: Lessons From Uganda.

Susan Reynolds Whyte And Michael A. Whyte. "The Values of Development Conceiving Growth And Progress in Bunyole" developing Uganda.

Thandika Mkandawire: "Thinking About Developmental States in Africa" UNCTAD, African Development in a Comparative Perspective, Study No 9.

Thandika Mkandawire And Charles C. Soludo. African Perspectives On Structural Adjustment: Our Continent Our Future.

Thandika Mkandawire And Charles C. Soludo. African Voices on Structural Adjustment.

Theodore H. Moran. Foreign Direct Investment And Development.

Thomas B. Gold. State And Society in The Taiwan Miracle.

Thomas G. Parry. "The Role of Foreign Capital in East Asian Industrialization, Growth And Development", Achieving Industrialization in East Asia.

Thomas M. Callaghy. "Lost Between State And Market: The Politics Of Economic Adjustment in Ghana, Zambia And Nigeria" Economic Crisis And Policy Choice: The Politics of Adjustment in the Third World.

Thomas M. Callaghy And John Ravenhill. "How Hemmed in? Lessons And Prospects of Africa's Responses to The Decline," Hemmed in: Responses to Africa's Economic Decline.

Thomas P. Tomich, Peter Kilby And Bruce F. Johnston. Transforming Agrarian Economies-Opportunities Seized, Opportunities Missed.

Timothy Longman. "Rwanda: Chaos From Above" The African State at a Critical Juncture: Between Disintegration And Reconfiguration.

T.H. Shen. The Sino-American Joint Commission on Rural Reconstruction: Twenty Years of

Cooperation For Agricultural Development.

Tekatoshi Ito. "Growth, Crisis And The Future of Economic Recovery in East Asia" Rethinking The East Asian Miracle.

U.K. Nyanteng And A Wayo Seini. "Agricultural Policy And The Impact on Growth And Productivity 1970-95" Economic Reforms in Ghana: The Miracle And the Mirage.

Vito Tanzi. "The Consistency Between Long-term Development Objective And Short-term Policy Instruments in Fund Activities" From Adjustment to development in Africa.

William Easterly. The Elusive Quest For Growth, Economists Adventures And Misadventures in the Tropics.

World Bank Accelerated Development in Sub-Saharan Africa: An Agenda for Action.

World Bank. Adjustment in Africa, Reforms, Results And The Road Ahead.

World Bank. African Development Indicators 2002.

World Bank. Building Institutions for Markets: World Development Report 2002.

World Bank. Can Africa Claim The 21st Century.

World bank. Finance For Growth: Policy Choices in a Volatile World.

World Bank. Sub-Saharan Africa: From Crisis to Sustainable Growth: A Long-term Perspective Study.

World Bank. The East Asian Miracle: Economic Growth And Public Policy.

World Bank. The State in a Changing World, World Development Report 1997.

World Bank. World Development Report 2002.

Yaw Asante, Frederick Nixon And Kwaku Tsikata. "The Industrial Sector And Economic

Development" Economic Reforms in Ghana: The Miracle And The Mirage.

Yoweri Kaguta Museveni. Sowing The Mustard Seed: The Struggle for Freedom And Democracy in Uganda.

Yung Chul Park. "Concepts And Issues" The Financial Development of Japan, Korea And Taiwan, Growth, Repression And Liberalization.

Yung Whee Rhee, Bruce Ross Larson And Garry Purcell. Korea's Competitive Edge: Managing The Entry Into World Markets.

Yusuf Bangura And Peter Gibbon. "An Introduction to Some Conceptual And Empirical Issues" Authoritarianism, Democracy And Adjustment.